Contents lists available at ScienceDirect



Journal of Economic Behavior and Organization

journal homepage: www.elsevier.com/locate/jebo

Preventing criminal minds: Early education access and adult offending behavior^{*}



Economic Behavior & Organization

Zelda Brutti^{a,b,*}, Daniel Montolio^{a,b}

^a University of Barcelona, Department of Economics, Torre 6; Carrer John M. Keynes 1-11, 08034 Barcelona, Spain ^b IEB - Institut d'Economia de Barcelona; Carrer John M. Keynes, 1-11, 08034 Barcelona, Spain

ARTICLE INFO

Article history: Received 9 February 2021 Revised 24 July 2021 Accepted 25 August 2021 Available online 20 September 2021

JEL classification: 126 128 K14 K42

Keywords: Public preschool Early education Universal preschool Kindergarden Offending behavior Crime Noncognitive skills Spain

ABSTRACT

In this paper we estimate the impact of higher early preschool access on offending behavior later in time, exploiting a nationwide public preschool expansion that took place in Spain over the 1990s. This is the first study providing evidence on the relationship between preschool and crime using a large-scale, universal-access program rather than targeted initiatives reserved for disadvantaged children. Our identification strategy relies on the staggered implementation of the Spanish preschool expansion across regions and birth cohorts. We link the education data to a unique administrative crime dataset recording the offenses committed in the region of Catalonia over the period 2009–2014. On average, a 1 percentage point increase in preschool access among the age 0–3 population yields 1.6% fewer crime actions during youth and young adulthood. Larger effects on impulsivecrime categories provide suggestive evidence for non-cognitive-skills improvements being important drivers of our results.

> © 2021 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/)

"Educate the children and it won't be necessary to punish the men." – Pythagoras

E-mail address: zelda.brutti@ub.edu (Z. Brutti).

https://doi.org/10.1016/j.jebo.2021.08.035

0167-2681/© 2021 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/)

^{*} For helpful comments and suggestions we thank Julian Cristia, Jennifer Doleac, Emma Duchini, Christina Felfe, Olle Folke, François Gerard, Chloe Gibbs, Olivier Marie, Emily Nix, Natalia Nollenberger, Johanna Rickne, Luis E. Rojas, Matteo Sandi, the editorial team as well as a number of anonymous reviewers. We benefited from feedback and discussions at the Institut d'Economia de Barcelona, Universitat de Barcelona, Universitat Autónoma de Barcelona, and at several workshops and conferences. The authors acknowledge support from grant 2015ACUP 00037 from RECERCAIXA (Obra Social La Caixa) and PID2019-109813RB-100 from the Spanish Ministerio de la Ciencia, Innovación y Universidades. Daniel Montolio also acknowledges support from grant ECO2016-75912-R from the Spanish Ministerio de Economía y Competitividad. All errors remain our own.

^{*} Corresponding author at: Pompeu Fabra University, Department of Economics, Edifici Jaume I, Planta 1; Carrer Ramon Trias Fargas 25-27, 08005 Barcelona, Spain.

1. Introduction

Early childhood is recognized by various fields of research as a key stage in life, carrying the power to shape future outcomes through youth and well into adulthood. There is wide consensus about early childhood assessments being good *predictors* for educational achievements, labor market success and health outcomes.¹ Even more importantly, a growing number of studies shows that *policy interventions* on early education are able to produce significant impacts in the short and even in the long run, especially when disadvantaged children are targeted.² The majority of studies looks at cognitive skill formation and derived outcomes, while along the *non-cognitive* and *behavioral* dimensions, rigorous large-sample evidence is scarce and mixed, often due to the lack of high-quality data on outcomes (Duncan and Magnuson, 2013). However, pedagogical theory clearly indicates that pre-kindergarden circumstances and experiences are able to forge an adult's spheres of anxiety and self-control, self-esteem and motivation, socio-emotional intelligence and aggressive behavior.³ Moreover, evidence from psychology, criminology and economics suggests that reducing early externalizing behavior reduces crime (Heckman et al., 2013).

Building on these premises, our paper studies the causal relationship between the access to good-quality early childhood education and criminal propensity later in life, and is the first to do so looking at a universal public preschool program. Previous evidence is based on small-scale, highly selective initiatives – such as Perry, Abecedarian and CARE in the US, which count significant crime reduction among their benefits. These programs are characterized by high per-pupil costs and intensive interaction with participating families. It is far from obvious that a 'standard' preschool program, ran on nationwide scale and open to all households, would still yield appreciable benefits in terms of adult offending behavior.

Starting in the early 1990s, a Spanish national education reform boosted the availability of public preschool for 3 yearold children. Due to the autonomy enjoyed by regional governments on educational matters, the reform was implemented in a staggered manner, and substantial variation in public preschool access for 3-year olds arose both across birth cohorts and across Spanish regions. For crime outcomes, we employ an exclusive dataset of highly-detailed administrative data obtained from police forces, recording offenses committed in the region of Catalonia over the period 2009–2014. Catalonia is populous, wealthy and has an attractive labor market, so that its resident population includes internal migrants coming from all other Spanish regions. We construct offending rates between ages 15 and 30 by region of origin and birth cohort, and relate the variation in offenses to the variation in preschool enrolment rates characterizing each region and birth cohort back in the 1990s. We estimate the causal impact of early preschool access on crime prevalence through an intuitive and robust fixed effects model.

We find a negative impact of early public preschool access on crime rates in youth and young adulthood: a 1 percentage point increase in age 0–3 preschool access back in the early 1990s yields on average 1.6% fewer crime actions in the 2000s. These magnitudes are smaller than those identified by evaluations of the selective US preschool programs; however, our cost-benefit analysis suggests that the monetary value of cost-of-crime savings produced by the Spanish public preschool expansion exceeded expenses by multiple times.

Our high-quality crime dataset allows us to observe detailed information about the type of offense committed. Exploiting this data, in the second part of our analysis we assess the impact that the national preschool expansion has had on different types of crime separately, which helps us to shed light on the potential mechanisms at work behind the main results. The crime types on which we estimate the highest preschool impacts are those of more impulsive and imprudent nature – those crimes for which literature has found poor socio-emotional and behavioral skills to be important drivers. Our findings suggest that the non-cognitive skills channel plays an important role in explaining the reduction in crime which followed the increased public preschool access in Spain. These results are in line with the existing evidence from the US, which our paper extends to a context of large-scale and universal-access.

We test the exogeneity of the Spanish preschool expansion pattern with respect to our regression model, by showing the absence of correlations with relevant region-cohort characteristics and by conducting placebo tests. The latter prove that the relationship between early preschool access and crime exists only within correctly matched cohorts and regions, while it breaks down when associating access rates of only slightly older peers, different cohorts or different regions of origin. We provide statistical evidence to support the remaining assumptions behind our estimation strategy, including those on internal migration behavior and the crime-age distributions. We discuss the threats to our empirical setup, highlight its advantages with respect to existing work, and perform a set of robustness checks to corroborate our findings.

The remainder of this paper is structured as follows. Section 2 reviews the literature. Section 3 presents the institutional background on preschool education in Spain, and illustrates the public preschool expansion occurred in the 1990s. Section 4 presents the data employed and discusses the characteristics of our sample of analysis. Section 5 explains our empirical strategy. Section 6 presents our main results, discusses potential channels behind these and illustrates our costbenefit exercise. Section 7 reviews the main assumptions underlying our empirical strategy and offers empirical evidence to back these up. Section 8 provides further considerations about strengths and shortcomings of our analysis, and concludes.

¹ Among others, see Burchinal et al. (2010), Nelson and Sheridan (2011), Le et al. (2005), Duncan et al. (2010), Nguyen et al. (2016), Case et al. (2002).
² Among others, see Currie and Thomas (1995), Garces et al. (2002), Ludwig and Miller, 2007, Magnuson et al. (2007), Conti et al. (2016), Campbell et al. (2002), Heckman (2006), Gertler et al. (2014), Conti and Heckman (2014).

³ Among others, see Gunnar and Barr (1998), Kagan and Snidman (1999), Moffitt et al. (2011), Maslow (1943), Almlund et al. (2011), Phillips and Shonkoff, 2000, Robinson (2007), Tremblay et al. (2004); Hahn et al. (2007).

2. Related literature

There is a small literature studying the impact of universal preschool programs on criminal behavior later in time. In fact, most studies on the effects of preschool education are either focused on the short-run, consider strictly cognitive outcomes or are based on highly selected populations. On the other hand, studies on the relationship between education and criminal behavior typically investigate teenage-year interventions rather than early childhood ones. The remainder of this section summarizes the main findings from recent literature related to ours.

Attending high quality pre-kindergarden programs is effective both towards increasing cognitive performance of children and towards improving their socio-emotional skills. The far-reach of the earliest educational choices into the adult socioemotional sphere has been recognized in economics only quite recently.⁴ Particularly influential in demonstrating such longrun impacts have been those studies evaluating high-quality US programs such as the Perry program, Project CARE and the Abecedarian project. Studies by Heckman et al. (2010a), Heckman et al. (2013) and Conti et al. (2016) follow individuals from age 3 up to age 40 and show that the Perry and Abecedarian programs had persistent positive impacts on academic motivation and externalizing behaviors such as aggressive, antisocial and rule-breaking conduct. They show that these short-run impacts in turn translated into significant improvements in a range of adult outcomes including marriage, health, healthy behaviors and, most importantly, a strongly reduced participation in crime, guantified in experiencing approximately half as many lifetime arrests as non-participants.⁵ These findings confirm those in the analysis by Schweinhart et al. (2005), which highlights crime prevention as the most beneficial lifetime impact of the Perry program. Long-run benefits from the Head Start program were documented using sibling-comparisons (Garces et al., 2002; Deming, 2009), variation in the program rollout across time and geography (Thompson, 2018) and combinations of the two strategies (Johnson and Jackson, 2019). Garces et al. (2002) find significant increases in educational attainment for both blacks and whites, as well as reduced probability of being charged with a crime for blacks; Deming (2009) however finds positive effects of Head Start on a range of young adult outcomes but not on criminal activity, which in his dataset is self-reported. Using more recent versions of the Head Start programs, Thompson (2018) confirms the positive impact on educational achievements and also documents increases in adult earnings and health, with larger treatment effects among blacks, children of lower-education parents, and children exposed to better funded programs. Johnson and Jackson (2019) again confirm increases in educational attainment and earnings for poor children, as well as reduced poverty and incarceration rates; they also show dynamic complementarities between exposure to the Head Start programs and later K-12 spending.

The common denominator of the research cited so far is its reliance on rather small and highly selected study populations. The access to the Head Start, Perry, Abecedarian, CARE and many other US state-level programs was and is reserved to individuals from low-income families and otherwise disadvantaged backgrounds. The majority of these programs includes mentoring for families, on top of the institutional education offered to children. It is far from obvious that the strong impacts on crime reduction identified in such contexts would hold if scaled up to encompass larger and more diversified populations, and ran on a standard, nationwide basis.

Most studies on universal or very-large-scale preschool initiatives have limited their attention to short-run behavioral outcomes or, when contemplating the long horizon, to strictly academic or labor market skills.⁶ An exception to the rule and close to our study is the recent contribution by Baker et al. (2019), who study the long-run effects of an universal preschool program introduced in Quebec starting in 1997 and exploit the fact that eligibility was staggered across birth cohorts. However, the Quebec program can be seen as an outlier in the preschool literature, since it has been found to be of particularly low cost and quality; earlier studies on its consequences had documented persistent negative shocks to noncognitive development of children and very little impact on cognitive skills (Baker et al., 2008; Kottelenberg and Lehrer, 2013). In fact, the recent long-run analysis shows persistence in the negative impacts on noncognitive outcomes throughout teenage years, and concludes that cohorts with increased child care access have worse health and lower life satisfaction later in life, as well as displaying higher crime rates between ages 12 and 20. Beyond the Quebec study, looking at impacts of *universal* preschool programs on criminal behavior there is a working paper by Smith (2015). The author focuses on a universal program started in Oklahoma in 1998 and leverages the birthdate cutoff for its first year of implementation. The sample is rather limited in terms of age, since it includes only individuals aged 18 and 19, but the results clearly point to very large reductions in criminal behavior among black children and no detectable impacts for white children. It is evident that further evidence on the impact of universal programs on criminal behavior is needed, and the Spanish preschool experience offers the characteristics of being nationwide, of universal access for all socioeconomic backgrounds, of good quality and sufficiently long-established to allow the analysis on crime outcomes for fifteen different birth cohorts.

The second body of literature enclosing our paper investigates the impact of education on criminal behavior in youth or adulthood. On this avenue, the vast majority of studies looks at the results of changing the minimum school leaving age,

⁴ The medical and pedagogical literature have been strongly supporting the effectiveness of school-based programs to reduce or prevent violent behavior for a longer time. For a systematic review, see Hahn et al. (2007).

⁵ These more recent studies refined and reinforced the findings by Heckman et al., 2009a,b, 2010, in which crime reduction was first identified as a major benefit of the Perry program, the main channel of the effect was enhancing non-cognitive and behavioral skills. Some years earlier, Campbell et al. (2002, 2008) had found that both Project CARE and the Abecedarian Project induced reduced marijuana consumption. However, at odds with the other studies, males were actually found *more likely* to report breaking the law.

⁶ Appendix A provides a short review of studies in these two strands of literature.

or at the results of attending higher- vs. lower-quality schools. Appendix A provides a brief summary of relevant literature in this area. The effectiveness of large-scale education policies in influencing offending behavior is generally supported, but all based on initiatives targeted at individuals aged between 14 and 18, for whom the effects of the additional education received often mingles with the pure crime-incapacitating effect of spending more time at school. Our paper extends the literature on the impact of education policies on crime by looking at a large early-years intervention – and more specifically, at one changing the start of formal education rather than its end.

3. Institutional background

Up until the early 1990s, Spanish preschool was organized into 'kindergarden' for ages 0 to 3 and 'nursery school' for ages 4 to 5, and the vast majority of children was starting preschool education at age 4. The public offer of nursery schools was adequate, and enrolling a child into nursery school meant securing a spot for primary education in the same center (Berea, 1992). However, mainly due to funding restrictions, the offer of public kindergardens for children aged 3 and below was utterly deficient and unable to meet the demand of families.⁷ For this age group, offer of private institutions was somewhat larger than the public, but was unregulated, expensive, and lacking any quality scrutiny on standards or staff credentials. Moreover, both public and private kindergardens were viewed as mere 'day-custody' centers (*"guarderías"*), typically carrying very little educational content (Berea, 1992; González, 2004). Informal daycare arrangements through extended family members or friends were the only alternative resource for many working mothers (Del Campo Urbano, 1982).

In 1990 the educational system was deeply reformed through a new regulatory framework known as LOGSE – *Ley Orgánica General del Sistema Educativo* (Ley Orgánica 1/1990), which incorporated a significant increase in the importance given to preschool education. Preschool was restructured so that kindergarden was reduced to cover ages 0 to 2 and the start of nursery school was anticipated to age 3. These changes triggered a rapid expansion of early public preschool services and especially boosted enrollment of 3-year-old children, which rose from around 5% to around 65% between 1987 and 2000. This was an entirely supply-driven expansion, as documented by the many authors reporting the severity of excess demand for public preschool spots throughout the Spanish context over that period (Análisis de la demanda de servicios para la primera infancia, 1992; Felfe et al., 2015; González, 2004; Gonzälez and Quiroga, 2003; González and Vidal Torre, 2005, 2006). The reasons of the insufficient public provision was mainly due to chronic scarcity of funding (Calero and Bonal, 1999; Rambla and Bonal, 2000; Bonal et al., 2005). Based on this evidence, we will refrain from including any demand-side considerations in this analysis.

Figure 1 shows the evolution of national preschool enrollment rates over those 14 academic years, and distinguishes between 0–2 year-olds and 3-year-olds from 1993 onward, when age-specific data became available. There are two important takeaways from this figure: first, over this period the growth in preschool enrollment was entirely driven by the public sector, while the private offer remained virtually unchanged; second, the growth in preschool enrollment was almost entirely driven by 3-year olds. In fact, we will focus our empirical analysis solely on the public sector and overlook the 0–2 age group, as will be discussed later on.

Public preschool spots are open to all families and heavily subsidized by regional governments, so that parents are charged on average only around 12% of the total cost of a preschool spot⁸, with additional public subsidies available to low-income families, but without ever being completely free-of-charge.⁹ Both low and middle-income families are the most common users of public preschool centers. This pool of users diverges both in terms of size and in terms of characteristics from those populating the smaller-scale Perry, Abecedarian and CARE programs in the US, making the contribution of our analysis all the more innovative.

It is important to note that through the new legal framework, the role of preschool transitioned from the previous 'daycustody' perspective to a proper educational stage, with class sizes limited to 20 pupils and the requirement of university degrees for teachers. Importantly, the new Spanish preschool recognized the importance of non-cognitive skills for children's comfort and success in social environments, so that national guidelines put particular emphasis on non-cognitive skills development. Children were promoted in their self-awareness, personal autonomy, knowledge of the environment, communication and representation skills, social, affective and personal advancement.

3.1. Staggered expansion

The decentralization of educational competences from the national to the regional level started in Spain during the early 1980s and protracted itself until the late 1990s, going hand in hand with the wider political decentralization process occurring in those years. As the new educational regulations were approved in 1990, Spanish regions (*Comunidades Autónomas*)

⁷ Being voluntary, preschool ranked low in the priorities of the public education agenda. Funding was limited and directed primarily to compulsory schooling stages, and became especially scarce over the 70s and 80s decades, as the 1960s baby-boom led to an overstretch of national educational budgets.

⁸ Based on data from the region of Andalusia for the school year 1994 (Temporeros y educación, Temporeros y educación (1997))

⁹ Access to public spots is regulated by a point system, which prioritizes children with older siblings already attending the same education center, or whose parents live or work in its catchment area. Extra points are attributed for low income status and to large families.



Fig. 1. Early preschool enrollment rates in Spain, 1987–2000 Note: Early preschool enrollment rates in Spain, by school year. Source: Statistics of the Ministry of Education. Public preschool rates are solid and total (public + private) ones are dash-dotted. The upper two series trace age 3 enrollment, the middle two trace the ages 0–3 average enrollment, the lower two trace ages 0–2 average enrollment. Before 1993, enrollment rates were not collected separately by age by the Ministry of Education.

were in the process of taking over educational responsibilities.¹⁰ As a result of their decisional and budgetary autonomy, the timing of the expansion in the public supply of early preschool varied considerably across the eighteen Spanish regions; Fig. A.1 shows the regional evolution of early preschool enrollment rates between school years 1987 and 2000.

As in several other cases of uncoordinated reform rollouts, the Spanish preschool expansion was staggered across time and space but not the result of an actual random assignment procedure (among several others, see Blanden et al., 2016; Duflo, 2001; Felfe et al., 2015; Havnes and Mogstad, 2011; Hoynes et al., 2011; Johnson and Jackson, 2019). Several studies on the Spanish case have failed to identify any salient determinant explaining the timing in the adoption of the 1990 educational reform across regions, or justifying the territorial differences in public preschool supply (Bonal et al., 2005; González and Quiroga, 2003; Flaquer and Oliver, 2002). Cross-regional differences in policy adoption have been attributed to a complex blend of local wealth, political preferences, differences in local financing models, socio-economic structure, local history, demographic distribution and migration, but without identifying any salient factor among these (Bonal et al., 2005).

Given that the cross-regional and cross-temporal variation in preschool expansion underlies our identification strategy, it is important to discuss whether it might be spuriously related to our outcome variable, criminal behavior later in time. As a first step, through the analysis presented in Table A.1 we are able to confirm the absence of significant correlations between the regional pattern of preschool expansion and a range of variables proxying institutional, financial and socio-economic factors characterizing each Spanish region and birth cohort. Importantly, we also introduce proxies for criminal activity back in time to identify any spurious relationship between education and crime dynamics pre-dating the 1990s preschool expansion – and we find none. Despite these results, one might still be worried that something in the mix of regional characteristics might drive both preschool enrollment patterns and the later criminal propensity of individuals growing up there. We thus further validate the exogeneity of region-cohort specific preschool enrollment levels with respect to our model through the compelling placebo tests illustrated in Sections 6 and 7.

¹⁰ Overall, more than 50% of the public education expenditure was administered by regional governments at the end of the 1980s, and the figure would rise to above 80% over the 1990s (Calero and Bonal, 2004). On the other hand, the involvement of the more local level (provinces and municipalities) in educational matters is, in general, quite limited in Spain.

4. Data

Data on regional preschool enrollment stems from the records of the Spanish Ministry of Education¹¹, which we personally digitalized, and data on resident children is provided by population series (by region of origin and year of birth) from the Spanish National Statistics Agency (INE).¹² We focus on birth cohorts 1984 to 1997, which are homogeneous in terms of educational characteristics beyond preschool access (most importantly, school leaving age) and are old enough to appear in our crime dataset later in time. Our main variable of interest is a measure of early preschool access, which we construct as follows:

$$P_{c,i} = \frac{\text{Preschoolers } (0-3)_{c+3,i}}{\text{Total children aged } (0-3)_{c+3,i}}$$

where *c* indexes the birth cohort and *i* indexes the Spanish region of origin. That is, we measure early preschool access for region *i* and cohort *c* as the number of 0–3 year olds attending public preschool in the academic year c + 3, over the total population of 0–3 year olds in the region. For example, the preschool access of a child born in the region of Andalusia in 1990 is measured as the number of preschool attendees among children aged 0–3 registered in Andalusia in the year 1993, over the total number of children aged 0–3 living in Andalusia in 1993. In effect, we use the full 14-year long enrollment data for the 0–3 age group but attribute all its dynamics to 3-year olds, our focus population. Figure 1 had highlighted the stylized fact that enrollment of 0–2-year olds remained almost stationary over our period of observation, and overlooking constant quantities shall not affect our estimates. However, this choice will imply the need to rescale our point estimates when interpreting them.¹³ Note that given the context of a constantly binding supply of public preschool spots, we read preschool *enrollment* rates as our preschool *access* measure.

For our criminal behavior outcomes, we use highly-detailed administrative data from police records, on offenses committed by identified authors in each municipality in the region of Catalonia over the years 2009–2014; Catalan police authority (*Mossos d'Esquadra*). Available information includes date and region of birth of the author, timing and location of the action, and a classification of the type of offense committed. We take the freedom of adopting the generic term "crime actions" to refer to these offenses, although these police records predate the judicial process that will confirm or lift the imputed charges. We construct our outcome variable, 'crime rate per 1000 inhabitants', as follows:

$$O_{c,i,m,y} = \frac{Offenses_{c,i,m,y}}{Residents_{c,i,m,y}} \cdot 1000$$

where $O_{c,i,m,y}$ measures offenses per 1000 inhabitants belonging to birth cohort *c*, born in region *i* and living in Catalan municipality *m* in year *y*. Resuming our previous example, we construct measures of crime rates among young adults born in the region of Andalusia in the year 1990 and living in the municipality of Barcelona in 2009, 2010, and so forth for all Catalan municipalities and years from 2009 to 2014.¹⁴ The descriptive statistics in Table A.2 and the regional data in Fig. A.2 show that the variance of crime rates is multiple times higher than its mean. This is due to a large mass of zeroes and a long right tail: that is, we observe many instances of zero crimes, and few instances of very high crime rates.

Our main sample consists of 153,924 observations, each referring to a tuple (c, i, m, y) – 'birth cohort c, born in region i, living in municipality m, in year y'. We observe the 10 birth cohorts between 1984 and 1993, born across 18 Spanish regions, living across the 948 municipalities in Catalonia over the 6-year period 2009–2014.¹⁵ Figure A.3 illustrates the composition of our sample indicating the share of observations by each region of origin and each birth cohort: as expected, the single most represented region of origin is Catalonia itself, but the remaining regions constitute a sample fraction going from around 60% for older cohorts to around 45% for younger ones.

5. Empirical strategy

In order to estimate the causal effect of early preschool on criminal behavior, we exploit the variation in preschool access both across regions of origin and across birth cohorts. Since each Spanish region expanded early public preschool following its own timeline, the preschool access faced by each individual at age 3 is given by the combination of her birth cohort and region of origin. Our identification strategy controls for systematic differences in preschool access across regions through

¹¹ "Estadística de la Enseñanza en España. Niveles de preescolar, general básica y enseñanzas medias", Ministerio de Educación y Ciencia, eds. 1987/88 - 2000/2001. We use public enrollment data for our main specifications, but also show results for total enrollment (public + private) in our robustness checks.

¹² Specific source: Municipal Register of Resident Population (Padró municipal d'habitants) 2000–2014.

¹³ Felfe et al. (2015) instead create age 3 enrollment rates by backwards imputation, starting in 1993. However, this method is based on making arbitrary assumptions about the year-by-year growth rate of enrollment, and since this represents an important portion of our identifying variation, we prefer to keep it as unaltered as possible.

¹⁴ This construction views offenders as part of the resident population in the municipality m in which they commit the offense, thus ignoring the possibility of having crimes "away from home". Criminology literature agrees that in the vast majority of instances, distance-to-crime is well below two miles (for a review of the evidence, see Ackerman and Rossmo, 2015). A more careful treatment of criminal mobility goes beyond the scope of this paper.

¹⁵ Not all cohort-region combinations are found in all municipalities and in all years: that is, in some municipality m and year y combinations, there are no residents born in region i in cohort c. These instances are excluded from the sample.

regional fixed effects, and for systematic differences in access across time through cohort fixed effects, so that only their combination is treated as exogenous.

We opt for a negative binomial (NB) regression model, as it is able to accommodate the over-dispersion and the count nature of our outcomes;¹⁶ in Appendix B we carry out a formal discussion of the NB model setup and show results on model fit to the data. Equation (1) represents our baseline model, which on the right hand side features early preschool access rates $P_{c,i}$, birth-cohort fixed effects C_c and region of origin fixed effects R_i :

$$O_{c,i,m,y} = \exp\left(\beta_0 + \beta P_{c,i} + \sum_c \gamma_{1c} C_c + \sum_i \gamma_{2i} R_i\right)$$
(1)

The indexes *c*, *i*, *m* and *y* respectively denote cohort, region of origin, municipality of residence and year of crime data. This simple baseline specification encloses the core of our empirical exercise: we estimate the relationship between the prevalence of criminal behavior among individuals belonging to a specific cohort and region of origin and their early preschool access rates, after accounting for birth cohort effects and region of origin effects. Our crime data allows us to carry out the analysis with high geographical detail and and yearly frequency, increasing our precision substantially with respect to a more aggregate level – as we show later in our robustness checks.

Departing from our baseline specification, Eq. (1), we gradually allow crime outcomes to depend on further elements. We introduce local fixed effects and local controls, to reduce potential bias stemming from characteristics of the local area in which crime actions are committed. We also introduce year fixed effects to reduce potential bias associated with the timing of crime actions. Finally, we introduce controls for economic conditions that individuals faced in their regions of origin during their preschool time, to alleviate concerns about time-and-region-specific factors correlating to preschool access and co-determining later criminal behavior. Our most complete specification, Eq. (2), can be illustrated as follows:

$$O_{c,i,m,y} = \exp\left(\beta_0 + \beta P_{c,i} + \sum_c \gamma_{1c} C_c + \sum_i \gamma_{2i} R_i + \sum_p \gamma_{3p} V_p + M_m \lambda + \sum_y \gamma_{5y} Y_y + Z_{c,i} \delta\right)$$
(2)

where V_p indicate province dummies, M_m indicates a vector of municipal control variables (unemployment rates, number of police officers per 1000 inhabitants, municipal revenues per capita, municipal expenditures per capita),¹⁷ Y_y indicate year dummies and $Z_{c,i}$ indicates a vector of cohort-region of origin controls (GDP per capita and unemployment rates). As will be illustrated in Section 6, our results remain considerably robust throughout these additions, corroborating the validity of our parsimonious baseline specification.

The main assumption underlying the empirical strategy just described is the exogeneity of the staggered preschool expansion pattern with respect to our preschool-crime relationship. As anticipated in Section 3.1, one might fear that our fixed effects and time-varying controls are not able to account for all relevant factors characterizing the region of origin and time period in which our 3-year olds grow up. More specifically, such institutional, economic or cultural circumstances might influence both the preschool access to which children are exposed and their criminal propensity later in time. We address this concern through a compelling placebo test, in which we replace preschool enrollment rates for 3-year olds with the contemporaneous preschool enrollment rates of 4 to 5 year-olds. The intuition is that one would expect any spurious factors driving age 3 enrollment rates as well as criminal behavior later in time to also drive age 4-5 enrollment rates - that is, the contemporaneous enrollment of only slightly older peers. Suppose we were to find a relationship between preschool access and later offending behavior only within the exact same birth cohort, but no relationship when using the access rates of slightly-older cohorts, which should be similarly influenced by any regional- and time-specific confounders. Such a result would provide strong support to the hypothesis that age 3 preschool access influenced children's adult offending behavior directly, and not because of spurious correlation due to contextual counfounders. The results of this test indeed confirm our prior and are shown alongside our main results in the next section. Among the robustness checks in Section 7 we show a further placebo which corroborates the fact that results do not hold when introducing mismatches between the cohort of treated children and the young adults whose crime outcomes are measured.

Further assumptions of our empirical analysis are the following: i) preschool access does not affect internal migration behavior in significant ways; ii) preschool access does not shift the *timing* of offending behavior in significant ways; iii) region of birth corresponds to region of residence at age 3; iv) private sector dynamics can be ignored. Section 7 discusses each of these assumptions in further detail and provides evidence to support them.

¹⁶ The more naive OLS option is not suitable for neither the count nature of the outcome variable, nor for its over-dispersion; the more traditional Poisson alternative with robust standard errors, instead, produces unbiased coefficient estimates but fits the data decidedly worse than the NB model. Notice that, in our setting, the large mass at zero does *not* call for a "zero-inflated" modeling strategy. The basic assumption behind such strategies is the presence of an 'excess' mass of zero-outcomes, due to the fact that part of the population does not participate in the count-generating process – leading to no chance of observing any outcome different from zero for that part of the population. In our case however, crimes could in principle have been recorded in any (*birth cohort, region of origin, municipality of residence, year of observation*) tuple, so we do not wish to impose zero-counts anywhere a priori.

¹⁷ Because of the large number of municipalities (over 900), we opt for introducing province dummies and municipal-level controls rather than municipality dummies. Following the suggestion of an anonymous reviewer, we will introduce further municipal-level controls in our robustness checks Section 7.1, in particular addressing the aspect of crime reporting rates and their relationship to local internal migration and social capital.

Table 1

The effect of early preschool access on offending rates.

	Negative Bind	omial				Poisson	OLS
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
A. MAIN RESULTS							
Public preschool rate (0-3)	-3.21***	-1.60***	-1.59***	-1.64***	-1.67**	-1.17*	-47.15*
	(0.67)	(0.60)	(0.58)	(0.63)	(0.66)	(0.63)	(25.59)
B. PLACEBO							
Public preschool rate (4–5)	-0.88***	-0.28	-0.22	-0.23	-0.28	-0.38*	-27.82*
	(0.31)	(0.20)	(0.18)	(0.20)	(0.20)	(0.20)	(15.20)
Test H_0 : $A_{\cdot} = B_{\cdot}$, p-value	0.00	0.02	0.02	0.03	0.04		
Local controls				Yes	Yes	Yes	Yes
Regional controls (at age 3)					Yes	Yes	Yes
Year of birth FE		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Region of birth FE		\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Province FE			\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Year FE			\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Mean(y)	29.69	29.69	29.69	29.69	29.69	29.69	29.69
sd(y)	160.02	160.02	160.02	160.02	160.02	160.02	160.02
N.obs	153,924	153,924	153,924	153,924	153,924	153,924	153,924
R-squared							0.01
Pseudo R-squared	0.00	0.00	0.00	0.00	0.00	0.06	
α (overdispersion if > 0)	36.29***	35.84***	35.70***	35.24***	35.24***		

Note: Each observation represents cohort 'c' born in region 'i' living in Catalan municipality 'm' in year 'y'. Local controls are measured in crime-observation years *y* and include municipal unemployment rates; municipal revenue and expenditure per capita; police officers per 1000 inhabitants. Regional controls at age 3 are measured in year c + 3 and include region of origin GDP per capita; region of origin unemployment rate. Results are not sensitive to changes in the variables used to proxy economic conditions at the region of origin during preschool age or in the Catalan municipalities during the crime-observation window. SEs clustered (1-way) by cohort-region in parentheses. * p < 0.10, *** p < 0.05, *** p < 0.01.

Throughout the analysis, we employ one-way clustering of standard errors at the cohort-region level (n = 252 clusters). This choice is mainly driven by the fact that we are not able to include cohort-region fixed effects to our model, since that is the level at which our treatment varies; we believe it is particularly important to adjust standard errors for the potential residual correlations between observations belonging to the same birth cohort and region of origin, i.e. the level at which the treatment is assigned (Abadie et al., 2017). The robustness of results to different inference options, in particular targeted at accounting for residual correlation between regions or cohorts, has been verified through additional background exercises not shown but available upon request.

6. Results

Table 1 shows the estimations we obtain from the models illustrated above.

Looking at the main results in Panel A, Column 1 shows that the correlation between early preschool exposure and crime rates is strongly negative: that is, higher levels of early preschool are on average associated with lower crime rates in the data. Column 2 shows the results of our baseline model (1), which represents our preferred specification. The following columns show the results of increasingly rich specifications; the estimation of the most complete model (2) is shown in Column 5. As we can see, the estimates for public preschool rates remain robust throughout the inclusion of various local and time fixed effects, as well as time-varying controls proxying for economic conditions at the region of origin and at the municipality in which crimes are committed. This corroborates our assumption that in the framework of our model, region-and-cohort-specific variation in public preschool rates can be considered exogenously assigned.¹⁸ Columns (6) and (7) illustrate the results of estimating our most complete specification through Poisson and Ordinary Least Squares instead of the Negative Binomial strategy.

Panel B shows the results of our placebo test. When we replace age 0–3 enrollment rates in year c + 3 with age 4–5 enrollment rates in year c + 3, the relationship with criminal behavior falls close to zero and loses any statistical significance. This result adds to the previous evidence that time-varying regional conditions characterizing children's early years are not explaining the relationship between preschool access and later criminal behavior. Preschool enrollment and crime are related only when they refer to the exact same birth cohort, while enrollment figures of slightly older peers – which we expect to be similarly influenced by any time-varying regional confounders – do not correlate with the outcome.

Our main results imply that, for the average birth cohort in our sample, a 1 percentage point increase in age 0–3 public preschool access turned into 1.6% fewer crime actions during young adulthood over the years 2009–2014, after taking into

¹⁸ Another way to express this is that, conditional on birth cohort and region of origin fixed effects, potential (or 'counterfactual') criminal behavior C^0 of each group of residents is mean-independent of the level of public preschool exposure it received: $E[C_{c,i}^0[C_c, R_i, P_{c,i}] = E[C_{c,i}^0[C_c, R_i], where <math>C_{c,i} = C_{c,i}^0$ if $P_{c,i} = 0$.

account region of origin, birth cohort, time and local fixed effects, as well as a battery of time-varying controls.¹⁹ However, an increase in enrollment among 0–3-year olds is in fact mostly driven by 3-year olds, as was discussed around Fig. 1. Now taking the simplifying assumption that enrollment among 0–2-year olds remained completely static over the period studied, we estimate that a 3 percentage point increase in public preschool access of 3-year olds yielded 1.6% fewer crime actions during young adulthood, on average.²⁰ That translates into a reduction of approximately 0.5 offenses per 1000 inhabitants, over a baseline mean of 29 offenses per 1000 inhabitants. We can also use our estimates to relate the cross-regional variation in enrollment to the cross-regional variation in offending rates: we find that the cross-regional variation in early-preschool enrollment rates during the years 1987–2000 explains 15% of the variation in offending behavior across regions of origin during the years 2009–2014.²¹

These impacts are quite large, but still substantially lower than those found by evaluations of the small-scale preschool programs in the US – a fact that may be explained by their exclusive focus on disadvantaged children, while our estimates reflect average effects across the whole population that benefited from higher preschool access. Also the large-scale studies estimating the effects on crime of changing minimum school-leaving ages typically find impacts that are up to five times larger with respect to ours (for a review of this literature, see Section A.2). Beyond considering shorter time horizons, crime reduction in those studies typically includes the pure incapacitation effect derived from spending time at school instead of on the streets.

6.1. Evidence from catalan cohorts

Following the suggestion of an anonymous reviewer, we have carried out an additional exercise to reinforce our main results. If the increased preschool access has a negative effect on crime propensity, one should be able to focus on the region of Catalonia – which is particularly well represented in our sample – and detect a difference in criminal propensity among adjacent cohorts of 3- and 4-year olds affected by the preschool expansion. There are two aspects to recall, in this context: the first is that there is no actual "before and after reform" in our data, given that the public early preschool expansion started in the early 1990s for Catalonia and protracted until the end of our observation window. Nevertheless, we can identify an "intensive expansion" period, covering the first half of the 1990s. The second aspect is that after the intensive preschool expansion started, *all* cohorts of children are started benefiting of higher early preschool access, although in different and increasing levels.

This said, the exercise we have carried out delivers results which confirm the priors stated above. Fig. A.4 shows the (standardized) differences in crime rates between individuals born one year apart: raw differences in Panel A and differences conditional on cohort and local characteristics in Panel B. As one would expect, during the 'flat' preschool access period (1980s), it does not matter whether an individual was born one year or the following, in terms of his/her crime outcomes later on. However, during the "intensive preschool expansion" period, being one year younger starts to matter, since younger individuals benefit of higher preschool access with respect to their 1-year-seniors: indeed, we can observe how during this period the crime propensity differences drop, implying lower crime for slightly younger individuals. The drop in the differences is more marked in the raw data than in the series adjusted for cohort and municipal characteristics, revealing that other institutional changes are likely to have contributed to an overall reduction in crime alongside the early preschool expansion. Finally, we can add that one would expect a deceleration in the difference between adjacent cohorts for the late-1990s, when the expansion slows down; unfortunately, these cohorts are too young to be included in our crime observation window and this hypothesis remains unverified for now.

6.2. Discussion on channels

Good quality preschool programs have been shown to improve both cognitive and non-cognitive or socio-emotional skills in children – and both of these channels may lead to a reduction in criminal activity. An improvement in cognitive skills lowers crime propensity through better economic opportunities and higher income from legal activities – as in the original model by Becker (1968). Improvements on the non-cognitive dimension, such as more self-control and lower tendency to aggressive behavior, reduce crime prevalence through criminal impulse moderation (among many others, see Brownfield and Sorenson, 1993; Burt et al., 2006; Burton Jr et al., 1998; Cauffman et al., 2005; Hirschi, 1969; Junger and Tremblay, 1999; Keane et al., 1993; White et al., 1994; Wikström and Svensson, 2010; Wikström and Treiber, 2007). Felfe et al. (2015) present evidence that the Spanish preschool expansion in the 1990s led to cognitive improvements at age 15, as measured by performance in PISA tests, but to the best of our knowledge, there are no studies measuring longer-run or non-cognitive impacts

¹⁹ Through additional exercises not shown but available upon request, we find that most of the effect appears to be driven by a reduction of offending intensity among values higher than 0, rather than by a complete eradication of crime in some (c, i, m, y) combinations. However, we are not able to shed light on the question of whether this is an intensive- or extensive-level reduction from the individual offenders' point of view.

²⁰ Given that 3-year olds represent approximately one third of the 0–3-year old population, any given increase in enrollment rates in the 0–3 group implies an approximately 3 times larger increase in enrollment of 3-year olds.

 $^{^{21}}$ Over the period 1987–2000, the standard deviation of age 3 public preschool enrollment across regions was 13 percentage points; and 3 percentage points for 0–3 enrollment rates. Over the period 2009–2014 the standard deviation in offending behavior across regions of origin was 10.8 offenses /1000pax. Our estimates imply that increasing age 0–3 public preschool enrollment by 1 cross-regional standard deviation decreases offending rates by 15% of a cross-regional standard deviation approximately.

in this context. In our research setting we are not able to directly check intermediate outcomes such as behavioral improvements among young adults, a method which has been employed by earlier studies to test the channels behind the effects of preschool on crime. Instead, we propose an alternative and novel way to obtain a better understanding of the potential mechanisms at play. Our unusually detailed dataset allows us to construct different offense categories and to estimate our empirical model separately by each of these: heterogeneities in impacts may provide suggestive evidence on the channel leading from more preschool to reduced criminal behavior. Our categories, whose more detailed descriptives can be found in Table A.3, broadly follow the aggregates used in police records, with some rearrangement dictated by criminology literature. There are so-called "economic motivation" crimes (44.61%), which basically include thefts and robberies; the category of "damages to property" (4.87%) and the category "fraud and falsity" (4.54%). These first three groups roughly correspond to crimes whose underlying motivation is appropriation of material wealth and whose nature is rather premeditated. Further categories are "pure violence" (18.71%) against other individuals, such as injuries and threats; "drugs/addictions" (2.59%); "sexual" crimes (0.57%) and "rules compliance", such as violations of parole, of public order or decency and offenses to public officials (17.91%). Moreover, there are crimes committed within family boundaries, separated into "gender violence" (1.27%) and the more generic offense "against the family" (4.94%), which include non-sexual mistreatments of family members as well as failure to provide economic support or adequate child care. These offense categories tend to be less planned and rather driven by impulsive behavior, and in this sense they feature a tighter relation with the non-cognitive skill spectrum.²²

Table 2 presents results by disaggregated categories of offenses. With few exceptions, the estimated impacts of increased preschool access are negative across all crime categories and model specifications. We observe that, overall, crime types which are relatable to non-cognitive skills show larger and more significant estimated impacts of preschool access, with drug crimes leading both in terms of size and significance of the effect. A one (three) percentage point increase in age 0–3 (age 3) preschool access reduces the number of drug crimes by between 8 and 9% in youth and young adulthood.²³ The categories of violent crimes and rules compliance also show particularly robust patterns of negative point estimates across all specifications, with magnitudes well above our average result.

Notably, our large-scale and universal-access results are completely aligned with those found on disadvantaged preschool program users in the US (Campbell et al., 2002; 2008; Heckman et al., 2010b; Campbell et al., 2014; Conti et al., 2016), where reductions in drug use and reduced participation in violent crimes are particularly salient among the estimated benefits. The influential paper by Heckman et al. (2013) proposes a decomposition of the effects of the Perry program into cognitive skill-and personality skill-related. The authors show negligible effects of program-induced increases in IQ in explaining treatment effects, among which important crime rate reductions. In fact, while IQ enhancements through the Perry program turned out short-termed, improvements in personality skills have been found persistent by earlier studies (Carneiro and Heckman, 2003; Heckman, 2000). The authors claim that the largest contributions to reduced criminal activity in young adulthood – as well as better labor market outcomes and health behaviors – are attributable to the substantial improvements in externalizing behaviors (aggressive, antisocial and rule-breaking behavior) induced by the preschool program. Our own results by offense categories also suggest prominence of the non-cognitive skills channel in explaining the reduction of adult criminal propensity. We infer that, like in several other international experiences, the Spanish public preschool expansion has succeeded in improving aspects such as socio-emotional wellbeing, self-control, aggressive behavior and anger management. In the long run, this translated into a particularly strong crime inhibition effect on the most impulsive offense categories.

As a final remark, we would like call attention to the fact that the early preschool effects that previous literature and ourselves have identified are not necessarily an exclusively *direct* effect of the additional formal education received.²⁴ It may well be that part of the effects of preschool expansion operate through extra-school channels, such as family resources and parental time, which are potentially affected by the expansion. For instance, one plausible story is that parents and especially mothers would be able to dedicate more hours to work due to the higher availability of preschool, so that children would benefit of a wealthier family background, which in turn would have a positive impact in cognitive and non-cognitive development and thus reduce offending behavior. On the other hand, this relationship has been found not to hold in all countries and settings (Havnes and Mogstad, 2011a), so that it would be interesting to analyze it for the Spanish case. It is also plausible that most young children would be exposed to a lower adult-to-child ratio in preschool with respect to a family-care arrangement, which has been shown to have detrimental effects on specific subgroups of children (Fort et al. (2020)). Exploring the contribution of these and other extra-school channels to the crime-reducing effects we find in our analysis would constitute a promising avenue for future research.

 $^{^{22}}$ It is worth noting that crime actions might be driven by multiple or mixed motivations. For example, not paying spousal support to the divorced partner – which in our classification is part of the offenses against the family – may be driven by financial reasons or be an emotional reaction to the separation. However, criminology literature supports the fact that specific offense categories – such as violence outbursts, drug use or sexual assaults – are more relatable to the non-cognitive skill spectrum than others – such as fraud, theft and robbery. We ignore three types of crime that occur very rarely in our Catalan dataset: crimes against the environment (0.32%), murder (0.27%) and arson (0.06%). Beyond their low counts, it is difficult to categorize these items across our groups.

²³ It is important to note that the Catalonian drug offense data we employ is mostly related to consumption and petty dealing tied to consumption, and that this does not include organized crime distributing drugs on a large scale.

²⁴ We thank an anonymous reviewer for pointing out this aspect.

Table 2

Impacts on disaggregated offense categories.

	Negative Binomial				
	(1)	(2)	(3)	(4)	(5)
Economic motivation					
PPR (0-3)	-2.33***	-1.58**	-1.62**	-1.41	-0.62
	(0.35)	(0.77)	(0.79)	(0.93)	(0.99)
Damages to property					
PPR (0-3)	0.54	1.82	1.81	1.71	-0.06
	(0.56)	(1.57)	(1.65)	(1.93)	(2.33)
Fraud and falsity					
PPR (0-3)	-6.77***	-3.21**	-2.64	-3.26*	-3.39
	(0.66)	(1.61)	(1.65)	(1.87)	(2.36)
Pure violence					
PPR (0-3)	-2.86***	-1.58	-1.86*	-2.76**	-2.32*
	(0.33)	(0.98)	(1.00)	(1.17)	(1.36)
Rules compliance					
PPR (0-3)	-4.57***	-1.36	-1.13	-0.72	-2.37*
	(0.33)	(1.10)	(1.06)	(1.07)	(1.32)
Drugs					
PPR (0-3)	-3.22***	-6.98**	-7.65***	-8.22***	-9.15**
	(0.92)	(2.94)	(2.91)	(3.06)	(3.71)
Against the family					
PPR (0-3)	-5.79***	-1.43	-1.08	-2.22	-0.41
	(0.50)	(1.64)	(1.72)	(2.22)	(2.70)
Gender violence					
PPR (0-3)	-5.13***	4.34	2.76	0.54	-1.47
	(1.03)	(2.74)	(3.15)	(3.36)	(3.63)
Sexual crimes					
PPR (0-3)	1.87	0.87	0.10	4.43	-5.36
	(1.96)	(4.82)	(5.09)	(5.44)	(6.36)
Local controls				Yes	Yes
Regional controls (at age 3)					Yes
Year of birth FE		\checkmark	\checkmark	\checkmark	\checkmark
Region of birth FE		\checkmark	\checkmark	\checkmark	\checkmark
Province FE			\checkmark	\checkmark	\checkmark
Year FE			\checkmark	\checkmark	\checkmark
N.obs	153,924	153,924	153,924	153,924	153,924

Note: The table reports early preschool exposure coefficient estimates for different types of offenses. Each cell of the table comes from a separate regression: each line refers to a different sample of a specific offense category; each column represents a different model specification employed, replicating those of our main results Table 1. SEs clustered (1-way) by cohort-region in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01.

6.3. Benefit-to-cost calculations

Our disaggregated results allow us to provide an approximate cost-benefit analysis of the Spanish preschool expansion, in which we compare the economic value of crime reduction to the cost of the expansion. Being limited to the sole crime dimension, our exercise is much less comprehensive than the one performed by Heckman et al. (2010b) for the Perry Preschool Program, which considers a range of benefits to society beyond crime reduction – such as better education, employment, earnings and health. Further, rather than addressing the full, lifetime rates of return of the program, we limit the exercise to calculating the average annual benefit-to-cost ratio over our period of observation (2009–2014).²⁵ We recover the unit cost of crime from recent international data, for two thirds of our offenses. We obtain an assessment of the annual monetary worth of preschool-induced crime reduction, using the point estimates from our main results. We then relate those benefits to the cost of the expansion program, based on unit costs of one extra preschool spot in Spain. In Appendix C we illustrate the precise data used and the calculations performed.

For the case of Catalonia in the year 2000, a 1 percentage point increase in public preschool enrollment rate would imply increasing age 0–3 enrollment rates from 22.3% to 23.3%, which translates into 4710 new spots, with an associated yearly cost of EUR 26.8 million (approx. 0.1% of the regional budget). The lower-bound estimate of associated benefits from crime reduction is EUR 113.9 million, which yields a benefit-to-cost ratio of around 4.25 at the very least.²⁶ One could further decompose that ratio distinguishing between "public" and "private" costs of crime. Looking at the reduction in public costs

²⁵ Calculating lifetime rates of return would imply the need to compute present values of *lifetime* costs and benefits of preschool expansion. In our data we observe crime committed by individuals aged 15–30: even though it corresponds to the age range where crime generally peaks, it still represents an incomplete share of the potential criminal life on an individual.

 $^{^{26}}$ The figure represents a lower bound estimate because it ignores around one third of offense on the benefits side, and because we use the more conservative crime cost estimates when several ones are possible. Through simple imputation of our average unit cost to the offense for which we are missing this information, our benefit-to-cost ratio would raise to 6.33.

only (EUR 37.1 million), we estimate the benefit-to-cost ratio at around 1.4. Looking at the reduction in private costs only (EUR 76.8 million), the ratio would be around 2.9. As a comparison, for the US Perry Program, Heckman et al. (2010b) report a benefit-to-cost ratio of 7.3 only in terms of crime reduction,²⁷ while Rolnick and Grunewald (2003) obtain a ratio of 5.7 for the same program.

Although recognizing the limitations of the simplified approach taken in this exercise, we believe it helps us providing a useful ballpark estimate of the economic advantages brought about by the Spanish preschool expansion. Considering nothing else than the advantages in terms of crime reduction, the monetary value of benefits to society as a whole exceed costs fourfold at the very least, and is well worthwhile even with reference to the 'public budget' only.

7. Assumptions and robustness

7.1. Internal migration

7.1.1. Effects on sample composition

The fact that we only observe offenses committed in the region of Catalonia implies that our crime rates by region of origin are based on the population of internal migrants – i.e. those individuals that have migrated from their region of birth to Catalonia during their youth or adulthood. The late 1980s and 1990s were periods of intense internal migration for Spain, due to the economic upturn and associated social dynamism the country witnessed, so that at the beginning of the 2000s almost a quarter of the population was residing in a region different from their birth (Romero-Valiente, 2003). Besides labor market incentives, migration was driven by residential preferences and the last return flows following the end of Franco's regime. During those years, Catalan provinces registered positive net migration flows which were above the national average although not exceptional. There are two aspects to internal migration that might affect the interpretation of our estimates. First, if preschool access affects internal migration decisions as well as offending behavior, we might run into an issue of sample selection. Specifically, our negative impacts on crime counts might reflect changes in the sample of Catalan residents we observe, rather than true reductions in offending propensity. Second, internal migration might jeopardize external validity, if internal migrants are intrinsically different with respect to the general population. We address these concerns by investigating whether our preschool access measure has any predictive power on the quantity or quality of internal migrants that we observe in Catalonia, and how the observables of internal migrants compare to non-migrants.

Internal migrants are individuals belonging to our birth cohorts (1984–1997) that reside in Catalonia during our observation window (2009–2014) and were born in a different Spanish region. We address the migration *quantity* aspect in Table A.4, which shows OLS regressions of the share of internal migrants on potential migration determinants, including preschool exposure. After accounting for cohort and region of birth fixed effects, we do not find any evidence for preschool exposure affecting the probability of migrating to Catalonia. Second, we look at the *quality* of internal migration. We identified our birth cohorts in 2011 Census data²⁸ and we classified individuals into migrants or stayers according to their residence and birth information. We look at individual characteristics of these two groups, such as education level and labor market activity, and cross-tabulate these against quintiles of preschool access. We would be worried if higher/lower preschool access were predictive of a gap in characteristics opening between migrants and stayers, as this would indicate an effect of preschool access on internal migration behavior. Fig. A.5 illustrates our exercise graphically and Table A.5 shows the detailed figures underlying the graphs. Differences in education or labor market activity between internal migrants and stayers seem i) not very large in general, and ii) not to exhibit trends or specific regularities across the preschool access on migration. Fact i) reduces external validity concerns; fact ii) fails to reveal any systematic effects of preschool access on migration choices, in line with our previous findings on migration quantity.²⁹ In conclusion, our results indicate that despite remaining a limitation of our empirical setup, migration-induced selection appears to be mild in our context of study.

7.1.2. Timing of migration

A further aspect to the migration to Catalonia is that its timing may be heterogeneous across individuals and regions of origin, occurring during early childhood for some and during young adulthood for others. While this is harmless if following a random pattern, it would be a source of concern if there are systematic features to this heterogeneity, which might correlate with unobserved confounders characterizing regions of origin. In particular, one could be worried that the environment faced at the region of origin *beyond age 3* continues to play a relevant role towards shaping criminal behavior later in time.

We do not have information on the year in which individuals moved from their regions of origin to Catalonia. However, we are able to mitigate the above-mentioned concerns by adding controls specific to each region-of-origin and birth cohort

²⁷ See their Table 6. The case that best approximates our scenario is the '0% discount rate' and Low Murder Costs. In fact, our analysis does not include murders, whose costs to society are estimated to be very high in their analysis. Excluding murders further reinforces the lower-bound nature of our benefit-to-cost estimate.

²⁸ We employ the 5-percent Census microdata sample available for public use (from the website of the National Statistics Agency, www.ine.es.)

²⁹ In a similar exercise shown in Table A.6 we perform logit and ordered logit regressions of migrants' observables on preschool exposure, cohort and birth region fixed effects. This strategy also fails to spot any correlation between migrants' characteristics and preschool. We repeat the quintile-based exercise for internal migrants to Catalonia only, which yields noisier results due to very small sample sizes for migrants. The graphical results are reported in Fig. A.6.

for ages beyond 3. In Table A.9 we show the results of augmenting the vector of region-cohort specific controls $Z_{c,i}$ in our most complete specification, Eq. (2), with control variables spanning further childhood years (row C1, ages until 12) as well as adolescence years (row C2, ages until 18). In fact the estimate of our coefficient of interest becomes somewhat larger and stronger with respect to our main results, indicating that if anything, the bias stemming from non-random timing of internal migration to Catalonia may be downplaying the effect of early preschool on offending behavior in our more parsimonious specification, which only includes controls for conditions faced at age 3.

7.1.3. Local social capital

Internal migration may directly or indirectly affect the level of social capital and institutional trust recorded at the local level.³⁰ This aspect deserves discussion in our setting, given that it is plausible to think that crime reporting rates may vary with local social capital and trust and thus be connected to the local internal migration rates that provide the base for our identifying variation.

If local immigration rates change local crime reporting rates because of the lower integration, different trust in institutions, different culture or different social capital characterizing internal migrants, other variables capturing local social capital and institutional trust, should be affected in a similar way as reporting rates.

The two variables we use to proxy local social capital and institutional trust are local association formation and local election turnouts, both of which have been extensively endorsed by literature on this topic. Associations are legal entities registered following The Catalan Associations Act, and they are defined as non-profit organizations, voluntarily formed by three or more individuals to serve a general or specific interest, through the sharing of personal resources, temporarily or indefinitely. There are different types of associations based on their main activities (e.g., cultural, civic rights, scientific knowledge, health and social welfare, etc.). Very common are those that promote civic rights in a particular place - typically, neighborhood associations.

We gather information on the association density per municipality and year and we compiled election turnout shares over various elections held over our observation window (2009–2014). We use standardized turnout rates at municipal level for two regional elections (2010 and 2012), two European elections (2009 and 2014) and one general election (2011); for the year 2013 we have imputed the values of the regional election held in 2012.

We re-estimate our models including the new local social capital proxies among our municipal controls and show the results Table A.9 (row C3). Reassuringly, our main results remain robust to the inclusion of the new social capital / institutional trust proxies and, if anything, the point estimates of the effect of early preschool access increase slightly.

7.2. Crime-age distributions

The interpretation of our main results may change if preschool merely changed the *timing* or the *black* age dispersion of offending behavior among youths, rather than its intensity. Similarly to the concerns about migration behavior, if preschool affects the timing of crime, our results might reflect a change in the observed sample rather than a true treatment effect. Through the regression analysis shown in Table A.7 we verify whether characteristics of the crime-age distributions are correlated with our preschool access measure. The dependent variables are peak crime age, low crime age, peak-to-low ratios, standard deviation and kurtosis of the crime-age distribution; independent variables are preschool access and the usual cohort and region fixed effects. We do not find preschool to carry any predictive power on the shape characteristics of observed crime-age distributions, further mitigating concerns about compositional effects underlying our results.

7.3. Region of birth and region of residence

Given that both our data on offenders and our data on population residing in Catalonia provide region of birth but not region of residence at age three, we assume that individuals were still residing in their region of birth at age three, and were thus exposed to the preschool expansion pattern that occurred in that region. This assumption, and in general the assumption of negligible mobility of children across the geographical areas at which analysis is conducted, is very common in papers with a structure similar to ours (including Blanden et al., 2016; Duflo, 2001; Felfe et al., 2015).

Those individuals who were *not* residing in their region of birth at age three are allocated a wrong preschool access level in our analysis. If completely at random, the presence of such mismatches weakens any true relationship that there might be between outcome (crime) and predictor (preschool); nevertheless, a non-random misallocation could potentially bias our results in any direction. Mobility of children under the age of 3 is in fact quite low across Spanish regions, as we can observe in Table A.8, however we correct for it following the weighting method by Card and Krueger (1992). We construct migration probability matrices between Spanish regions, using cohort and age-specific migration data from residence variation records kept by the Spanish National Statistics Agency (*Encuesta de Variaciones Residenciales* 1988–2000, INE).³¹ For each birth cohort *c*, the matrix P^c cross-tabulates region of birth (*i*) and region of residence at age three (*j*), and its elements p_{ij}^c indicate the

³⁰ Thanks to an anonymous reviewer for bringing up this point.

 $^{^{31}}$ Complete yearly migration data is available for birth cohorts 1988 onward. Migration records for earlier cohorts are missing for ages 0 (1987), 0–1 (1986), 0–2 (1985) and 0–3 (1984). To the missing records, we impute the age-specific migration rates of the immediately younger cohort – which are likely to represent slight over-estimates of the true rates, given that mobility trends are increasing over time.

probability of residing in any region j at age three, while being born in region i. Based on these probabilities, for each birth cohort c and region of origin i, we construct mobility-weighted preschool access measures $P_{c,i}^*$ as follows:

$$P_{c,i}^* = \sum p_{j,i}^c P_{c,j}$$

where $P_{c,j}$ are the age three preschool exposure rates for cohort *c* in region *j*. We then re-estimate our main regression models using $P_{c,i}^*$ instead of the original, unweighted $P_{c,i}$, and report results in Table A.9, row C4. As we can see, estimates show only marginal changes with respect to the original ones, confirming the inconsequential role played by preschool-age mobility in our context.

7.4. Private sector

We address any residual skepticism about the potential influence of the sluggish private sector dynamics mentioned earlier, when describing Fig. 1. Given the characteristics of our empirical model, and that it was the public sector expansion driving the variation in our data, we do not expect that ignoring the private sector is relevant to our results. We estimate our models using total preschool access (public and private), instead of public only. Indeed, as we see in Table A.9, row C5, coefficients slightly decrease due to the higher mean of the new predictor variable, but all result patterns remain unaltered.

7.5. Additional robustness and placebo

We test whether our conclusions are sensitive to a refinement to our Negative Binomial model, that is accounting for different amounts of exposure. The likelihood of observing offending behavior might change across observation units, depending on the amount of population represented by each of them: in our dataset, we are more likely to observe criminal behavior on larger resident groups than in less-represented groups, for any given underlying population behavior. In Table A.9, row C6, we thus show results of our exposure-augmented model, in which the exposure variable is given by log of the number of residents for each cohort-region-municipality-year combination. In fact, the estimated impact of preschool access on offending behavior is actually somewhat larger once exposure is accounted for, with respect to our more conservative main specification.

In the following rows of Table A.9 we test the robustness of our results against the exclusion of specific regions from our dataset. In row C7 we exclude regions which are islands or territories (Canary Islands, Balearic Islands, Ceuta and Melilla): beyond being both culturally and institutionally different from the Spanish mainland in several aspects, they are small and present the highest variability in offending behavior (see Fig. A.2). In row C8 we exclude the two largest regions in our sample, Catalonia and Andalusia, to verify that our results do not hinge exclusively on these. In row C9 we exclude those regions which present irregular enrolment rate patterns, potentially due to missing data or misreporting issues (Comunidad de Valencia, Galicia; see Fig. A.1). For all three sample restrictions, the estimated relationship between preschool access and crime remains negative and sizeable; however when excluding the two largest regions we lose statistical significance as well as around 30% of magnitude.

We allow for changes in the level of geographical aggregation of our data, in order to rule out any spurious correlations specific to the municipal level of analysis. Row C10 and C11 in Table A.9 show results at the provincial and regional level respectively. Crime rates by birth cohort and region of origin, as well as local economic controls, are now computed at these two higher geographical levels (see descriptives in Table A.10). Point estimates remain in the same ballpark as the original ones at the municipality level, but standard errors inflate. These checks confirm the absence of any spurious correlations specific to our municipal level of analysis. At the same time, they reiterate the remarkable value of the unusual level of geographical detail in the crime data we are able to employ, which allows to pin down effects with greater precision.

Finally, we conduct a second placebo test to remove concerns of accidental correlation between preschool access (*P*) and offending behavior (*O*) across time or across space. We repeatedly simulate results from our model after replacing the true birth cohort or region of origin with random different ones. The intuition is that if no genuine relationship between *P* and *O* existed and our main results were the result of chance, we should be able to obtain similar estimations from a portion of those simulations. Fig. A.7 shows the outcomes of estimating Eq. (1) one-hundred times, with a placebo cohort (Panel (a)) or a placebo region of origin (Panel (b)) assigned to each group of residents, and compares these to our original results. In both experiments, it is striking how our original estimates are far off any simulation obtained from fictitious birth cohorts or regions, if anything, estimates tend to reach statistical significance on the positive side of the spectrum. These results once again confirm that the negative and significant relationship between preschool access and later offending behavior exists only when cohorts and regions of origin are correctly matched between treated children and observed young adults.

8. Conclusions

Our analysis on the Spanish preschool expansion has shown that the effectiveness of good-quality preschool programs in reducing crime during youth and young adulthood scales up and holds even in the context of nationwide, universal-access programs. We find that on average, crime types which can be related to the non-cognitive skills sphere show the largest

Z. Brutti and D. Montolio

and most significant estimated impacts of preschool access, with drug crimes and pure violence leading both in terms of size and significance of the effect. We interpret this as evidence suggesting that the influence of preschool on the non-cognitive skill dimension is strong, persists into adulthood and is powerful enough to display effects on antisocial behaviors. Our estimated effect magnitudes are smaller with respect to those found on reduced samples and disadvantaged contexts in the US, but our back-of-the-envelope calculations suggest that the economic value of crime-reduction is still multiple times larger than the program cost.

Compared to the existing US analyses, our study of the Spanish context presents a few disadvantages. The probably most significant one is the unavailability of skill data for our population of children or young adults: in this sense, we are missing an empirical measurement of 'intermediate outcomes' between preschool access in early childhood and criminal activities in young adulthood (such as in Heckman et al., 2013). Being able to provide 'first-stage evidence' on cognitive and/or non-cognitive skills improvements among program participants would represent an obvious gain for this study. Alas, being able to track individual program participants and to test their skills is unusual in the context of universal-access programs, and more typical in contexts characterized by a restricted and supervised participation. On the other hand, we are proposing an outcome-driven approach to infer the importance of the cognitive versus non-cognitive skills channels in explaining effects on crime, taking advantage of the high quality crime data at our disposal. With respect to the analyses by Heckman and coauthors, and to other papers evaluating preschool programs in the same spirit, our crime categories are finer and more detailed, allowing to relate crimes to either skill group with some confidence. We view this as a novel and valuable contribution to the literature which, beyond being applied to the evaluation of the Spanish preschool experience, may prove useful as a research method in other similar contexts. Furthermore, despite the fact that our sample is the result of internal migration, we trust that our study offers a wider angle on the relationship between preschool and adult criminal behavior with respect to analyses based on small-sample and restricted-participation programs.

Finally, it is worth dedicating some thought to the distribution of our treatment effect across the population of Spanish children which was affected by the preschool expansion program. Existing literature strongly suggests that children raised in disadvantaged socio-economic backgrounds are those benefiting the most from the introduction of good quality preschool programs (Guryan et al., 2008; Kimmel and Connelly, 2007). Regrettably, the lack of socio-demographic information on offenders makes us unable to analyze heterogeneity in the effect across individual characteristics and empirically validate this prediction for the Spanish case. As a second-best attempt, in Table A.11 we show heterogeneity of the impact of preschool access on crime by relative economic advantage of entire birth cohorts. We find weak evidence for a larger effect for cohorts who found themselves in a relatively disadvantaged economic context at age 3, our indicators of disadvantage being a) below-median GDP per capita; b) above-median unemployment rates; c) hardest-hit by the 1993 Spanish economic crisis. However, these estimates are quite noisy, and unsurprisingly so, since the importance of the heterogeneity in average socio-economic background across cohorts does not compare to the heterogeneity across individual characteristics *within* cohorts. However, all things considered, there are reasons to trust that also in Spain the public preschool expansion program helped reducing inequalities in society by benefiting disadvantaged children more than those from wealthier families, with sizeable repercussions on offending behavior in youth and young adulthood.

Declaration of Competing Interest

I gratefully acknowledge financial support from grant ECO2016-75912-R from the Spanish Ministry of Economy and Competitiveness, and from grant 2015ACUP00037 from RECERCAIXA (Obra Social La Caixa).

The authors acknowledge financial support from grant 2015ACUP 00037 from RECERCAIXA (Obra Social La Caixa) and PID2019-109813RB-I00 from the Spanish Ministerio de la Ciencia, Innovación y Universidades. Daniel Montolio also acknowledges support from grant ECO2016-75912-R from the Spanish Ministerio de Economía y Competitividad.

Appendix A. Literature review supplement

A1. Preschool education and adult behavior

We present a short review of studies on universal or very-large-scale preschool initiatives looking at short-run behavioral outcomes. In fact, even in the short-run, evidence on behavioral impacts of preschool obtained from large populations is discordant. Loeb et al. (2007) use data from the Early Childhood Longitudinal Study and conclude that center-based preschool education has overall negative effect on socio-behavioral measures at the start of kindergarden. On the contrary, Figlio and Roth (2009) use differences in access within a family and find that, in Florida, public pre-kindergarden participation reduces behavioral problems in elementary school – especially in disadvantaged neighborhoods. Also Berlinski et al. (2009) find positive results in Argentina, where third-graders improved attention, participation, discipline and effort following attendance of pre-primary education. Gupta and Simonsen (2010) find that in Denmark, compared to home care, enrollment in preschool programs at age three does not lead to any significant differences in children's outcomes at age 7. Felfe and Lalive (2018) find that public preschool rollout in one German state induced significant socio-emotional skill improvement measured at school-entry age, but especially so for children with a lower propensity to attend preschool in first place.

Next, we summarize research contemplating long-horizon effects of large-scale preschool initiatives, focusing on strictly academic or labor market skills. Several studies find strong cognitive-skill effects for disadvantaged groups but small or

negligible effects on the remaining population (Blanden et al., 2016). In Italy, Carta and Rizzica (2018) find no effects on school performance following a subsidized expansion of preschool places for 2-year olds. Blanden et al. (2016) exploit the staggered implementation of preschool for three-year olds across Local Education Authorities in England in the early 2000s, finding small improvements in attainment at age 5, but no apparent benefits by age 11, and attributing the failure to low quality of some preschool centers. Universal child care programs in Norway have been evaluated by Havnes and Mogstad (2011b, 2015), who find positive earning effects for children from low-income families but actual losses for upper-class children. Cornelissen et al. (2017) look at an universal preschool program in Germany and exploit its staggered implementation across municipalities, finding that school entry examination gains from the program were particularly high for children from disadvantaged backgrounds. In Europe, Dumas and Lefranc (2012) evaluate the effect of the expansion implemented in France over the 1960s and 70s, finding persistent and inequality-reducing effects on educational and labor market outcomes. Finally in Spain, Hidalgo-Hidalgo and García-Pérez (2012) find a positive impact of preschool on results in language, mathematics and reading using TIMSS-PIRLS data; Santín and Sicilia (2015) find a positive impact of more years of preschool on academic results at 4th grade; Felfe et al. (2015) look at the same public preschool expansion we exploit and find a positive impact on PISA reading and math scores, especially for girls and for students with low socio-economic background.

A2. Large-scale education policies and offending behavior

What follows is a brief summary of studies analyzing the impacts on offending behavior stemming from large-scale education policy initiatives, all of which were targeted to teenage populations. Lochner and Moretti (2004) show that between the 1960s and 1980s in the US, completing high school reduced the probability of incarceration by about.76 percentage points for whites and 3.4 percentage points for blacks, with the biggest impacts on murder, assault, and motor vehicle theft. Deming (2011) finds that black males who won access to their first-choice high school through a lottery commit 50% fewer crimes over the 7 years following assignment. Anderson (2014) finds that minimum dropout age requirements have a significant and negative effect on property and violent crime arrest rates for individuals 16 to 18 years old, between 1980 and 2008. He suggests the incapacitation effect to be the main driver, but notes that other channels such as human capital impacts cannot be ruled out. Again looking at changes in compulsory schooling laws in the US, Bell et al. (2016) conclude that credible causal estimates of the education-crime relationship cannot in general be identified for the more recent 1980-2010 period, though they can for some groups with lower average education levels (in particular, for blacks). Machin et al. (2011, 2012) focus on a 1972 change in the UK school leaving age and estimate that a 1% increase in the proportion of men staying on at school after the compulsory school leaving age reduces male youth crime by between 0.85 and 1.7%, with stronger effects on property than on violent crime. They also pin down significant impacts on other productivity-related economic variables such as gualification attainment and wages, demonstrating that the incapacitation effect of additional time spent in school is not the sole driver of the results. The paper by Buonanno and Leonida (2009) look at Italy over the period 1980–1995 and shows that graduating from high school, as well as attending more years of education in general, has a negative effect on crime rate over and above its effect through labour market returns. In the Netherlands, Groot and van den Brink (2010) show that the probability of committing crimes like shop lifting, vandalism and threat, assault and injury decreases with years of education – but find the opposite result for tax fraud.

Appendix B. Negative binomial modeling and fit to data

Here we illustrate our modeling strategy more formally, following Long and Freese (2014), Hilbe (2011) and Cameron and Trivedi (2010). In negative binomial regression, the outcome variable y_j is assumed to follow a Poisson distribution whose parameter μ_j^* – representing expected counts and their variance – is a function of a covariate vector \mathbf{x}_j , a parameter vector $\boldsymbol{\beta}$ and an unobserved component v_j . The unobserved component it is such that e^{v_j} follows a gamma distribution with mean 1 and variance α , where α is known as the over-dispersion or shape parameter.³² More formally, the NB regression model can be expressed as follows:

 $y_j \sim \text{Poisson}(\mu_j^*)$

where *j* indexes single observations, and

$$\mu_j^* = \exp(\mathbf{x}_j \boldsymbol{\beta} + \nu_j) = \exp(\mu_j + \nu_j)$$

and

$$e^{\nu_j} \sim \text{Gamma}(1/\alpha, \alpha)$$

This model allows for dispersion to depend on the expected mean of each observation. However, we do not allow the dispersion *parameter* α to vary across observations, that is, we are estimating a single α for our entire crime dataset.³³

³² The unobserved component v_j is what differentiates the NB model from the Poisson model, allowing for over-dispersion in observed outcomes by adding an observation-specific element. The higher α , the larger the over-dispersion; with $\alpha = 0$ we are back to the Poisson regression model.

 $^{^{33}}$ In our main analysis we are also abstracting from the so-called 'offset' variables, which can be included into the NB model to allow for outcome events y_i to be observed over different amounts of exposure, and whose coefficients are constrained to 1. We thus consider the amount of exposure to

We assess how well the baseline specification of our negative binomial (NB) model, Eq. (1), is able to fit the distribution of our crime data, and how it performs compared to the more traditional Poisson choice. As a first approach, Table A.12 shows descriptive statistics of the predicted distribution of count probabilities generated by the two modeling options, and compares it to the actually observed counts. While the *mean* of the count distribution is well approximated by both the NB and the Poisson models, the fit of the former is clearly superior to the fit of the latter when dispersion measures are taken into account. The large standard deviation, skewness and kurtosis of actual counts are very well replicated by our NB specification, while the Poisson model fails to do so.

Model fit for count models can also be assessed through a rootogram (Tukey, 1972; Tukey, 1977; Wainer, 1974; Kleiber and Zeileis, 2016), which compares the empirical distribution of a variable to a theoretical distribution. The two theoretical distributions we consider are the Poisson distribution and the NB distribution, characterized by our parameter estimates as in Eq. (1). Count residuals are computed as the difference between observed counts and the counts predicted by the theoretical model. The rootograms in Fig. A.8 plot the square root of these count residuals,³⁴,³⁵ with positive bars indicating underestimation of actual counts and negative bars indicating overestimation of actual counts. The Poisson specification (right panel) massively underestimates zero counts and considerably overestimates small counts. The NB model (left panel) performs significantly better, predicting zero counts quite well and over- or underestimating the remaining counts to a lesser degree with respect to the Poisson model. Finally, Table A.13 reports AIC and BIC for both model choices. Unsurprisingly, these criteria too favor using the NB model over a Poisson specification in our context.

Appendix C. Benefit-to-cost calculations

To the best of our knowledge, estimates for the social costs of crime are not available for Spain, and calculating these from scratch are beyond the scope of this paper. Instead, we make use of the estimates for the UK provided by Heeks et al. (2018), which should serve as a good proxy: these are estimates of how much a single crime action costs to society, including costs in anticipation of crime (defense expenditure and insurance administration), costs as a consequence of the crime (value of property stolen, physical and emotional harm, lost output, health services and victim services), and costs in response to crime (police costs and judiciary costs). These unit cost estimates are provided separately by crime type, and may be split into 'private' costs – associated with personal defense expenditure, insurance administration, value of property stolen, physical and emotional harm and lost output– and 'public' costs – associated with health services, victim services, police costs and judiciary costs. Matching the crime categories available in Heeks et al. (2018) to those present in our analysis,³⁶ we obtain unit cost estimates for around two thirds of the total offenses recorded in our sample, while the remaining third was not considered by Heeks et al. (2018).

We use our baseline preschool impact estimates to approximate the number of crimes avoided as a result of the preschool expansion,³⁷ limiting ourselves to those offenses we have cost estimates for. Lastly, we combine number of crimes avoided with unit costs and obtain our estimates of cost of crime savings - which we report in Table A.14. The table shows estimated total and public cost savings separately by crime type. Focusing on public unit costs of crime for thefts is 900.7€ (46.4% of total costs); robberies 7,669.8€ (48.1%); car thefts 5,629.2€ (38.9%); criminal damage 830.3€ (43.7%) and for fraud and falsity is 422.2€ (23.3%). Regarding non-cognitive crimes, for violent crimes we have injuries 4,813.0€ (24.3%); threats 3,293.1€ (39.5%); sexual and gender violence 2,181.3€ (23.8%). For sexual crimes we have used the conservative cost not including rapes whose estimated cost to society is six time higher.

In Table A.14, the Multiplier column shows the inflation factors we use to obtain estimates of the number of crimes actually occurred, based on our information on crimes reported and whose offender is known. Note that both the likelihood of being reported to police and of having the offender ascertained vary greatly by crime type. Using a second dataset for total reported crimes in Catalonia, we obtain a first inflation factor from crimes with known offender to crimes reported. Using multipliers calculated by Heeks et al. (2018) based on victimization surveys, we account for the difference between reported and real crime rates. In Table A.14, we show the final result of this two-step inflation procedure.

To complete our cost-benefit exercise, we compare the savings due to crime reductions with a cost estimate of the public preschool expansion. We take the average cost of a preschool post in 2004 in Catalonia from the analysis of Tarrach et al. (2011), who value it at 5,690.27 \in . This estimate is consistent with the reported costs in Madrid in 2013.³⁸ The main results of the benefit-to-cost estimations of a preschool expansion are reported in the main text, Section 6.3.

be homogeneous across our crime counts. In other robustness specifications (not shown, but available upon request), we introduce age as the exposure variable: individual age during our observation window 2009–2014 may determine the likelihood of being observed engaging in criminal behavior. Our results remain unvaried to such additions.

³⁴ The square root transformation prevents smaller counts to be obscured by larger counts in the graph.

³⁵ To improve readability, we show counts up to 100 crimes per 1000 inhabitants.

³⁶ We have matched what Heeks et al., 2018 define as "violence without injury" to our "threats". When computing the cost reduction impact of a preschool expansion we use for both injuries and threats the results we obtain for our pure violence category. Similarly, for thefts, robberies and car thefts we use the estimated results for our economic motivation category. Finally, we have assigned the unit costs of other sexual offenses to our gender violence typology of crime.

³⁷ We use our point estimates and ignore the uncertainty around it.

³⁸ This unit cost is consistent with that reported in the news for Madrid in 2013 of 5100€ see https://www.elmundo.es/elmundo/2013/08/27/madrid/ 1377605557.html (last accessed April 2020). Moreover informal reports from 2017 point out to a cost of 7000€ per year and pupil in public preschools

Appendix D. Figures



Fig. A.1. Early preschool (age 0-3) enrollment rates in each Spanish region, 1987–2000. *Note:* Age 0-3 enrollment rates by Spanish region and school year. Public preschool rates in solid black, private ones in solid grey and total ones dashed.

in Barcelona, see http://nadaesgratis.es/admin/tarifacion-social-de-las-guarderias (last accessed April 2020). As a reference, the estimated cost of the Perry Program per child in 2006 was \$17,759 (see Heckman et al. (2010b)).



Fig. A.2. Crime rates by regions of origin. Note: The figure plots means (black) and standard deviations (gray) of crime rates by region of origin. Dashed lines indicate overall mean and standard deviation.



Fig. A.3. Representation of birth cohorts and regions of origin in our sample. *Note:* The figure plots sample fractions by birth cohorts (x-axes) and by region of origin (y-axes). Overall percentage representation of each cohort and region is indicated in parentheses next to its label.





Fig. A.4. Differences in crime rates between adjacent cohorts of Catalans. *Note:* The figure scatters pairwise differences in crime rates between adjacent cohorts of Catalans, the younger cohort with respect to the older (black dots), averaged across all municipalities in our sample. Highlighted in gray is the period of most intensive public preschool expansion (solid line). Panel A shows raw differences per 1000 inhabitants; Panel B shows the residuals of the differences regressed on cohort and municipal characteristics. All pairwise differences have been standardized to the pre-expansion period average.



Fig. A.5. Characteristics of internal migrants by preschool exposure quintile. *Note:* For internal migrants and stayers, the figure plots shares of individuals by highest education level achieved (Lower Secondary, Higher Secondary, University or equivalent; all education level categories are mutually exclusive and the omitted category is "Less than secondary education"), activity status (working or studying versus neither) and part-time work (yes or no), in each quintile of preschool exposure (Q1 is lowest and Q5 is highest exposure). The difference in shares is dotted, and 95% confidence intervals are capped. Source: 2011 Census microdata (5-Percent Public Use Microdata Sample). Table A.5 shows the regressions on which the figure is drawn.



Fig. A.6. Characteristics of internal migrants (to Catalonia only) by preschool exposure quintile. *Note:* For internal migrants and stayers, the figure plots shares of individuals by highest education level achieved (Lower Secondary, Higher Secondary, University or equivalent; all education level categories are mutually exclusive and the omitted category is "Less than secondary education"), activity status (working or studying versus neither) and part-time work (yes or no), in each quintile of preschool exposure. The difference in shares is dotted, and 95% confidence intervals are capped. Source: 2011 Census microdata (5-Percent Public Use Microdata Sample). Missing data points are due to having fewer than 10 individuals in the migrants group.

(a) Random birth cohort assignment



(b) Random region of origin assignment



Fig. A.7. Random birth cohort or region of origin assignment. *Note:* The two panels plot early preschool exposure coefficient estimates and 95% confidence intervals obtained from our baseline model after randomizing birth cohorts (a) or region of origin (b) one-hundred times. In each sub-figure, the black marker at the top represents our original estimate.



Fig. A.8. Model Fit. Note: The figure plots differences between observed and predicted counts produced by our baseline Negative Binomial specification (left) and the alternative Poisson specification (right).

Appendix E. Tables

Table A.1

.

Determinants of regional public preschool enrollment rates (0-3), 1987-2000 .

	(1)	(2)	(3)	(4)	(5)
LOGSE reform	0.21***	0.19**	0.19**	0.19**	0.19**
	(0.05)	(0.08)	(0.08)	(0.08)	(0.08)
Public capital expenditures (% GDP)	0.08	0.08	0.08	0.09	0.08
	(0.06)	(0.06)	(0.06)	(0.06)	(0.06)
Stock of public capital in education pc	-0.09	-0.10	-0.10	-0.11	-0.13
	(0.12)	(0.11)	(0.11)	(0.10)	(0.11)
GDP pc		-0.00	-0.00	-0.00	-0.00
		(0.00)	(0.00)	(0.00)	(0.00)
Young unemployment/1000p		-0.02	-0.02	-0.02	-0.01
		(0.04)	(0.04)	(0.04)	(0.04)
Education level active population		-0.11	-0.12	-0.17	-0.19
		(0.18)	(0.18)	(0.21)	(0.20)
Female labor force participation			-0.02	-0.00	0.02
			(0.19)	(0.20)	(0.21)
Court activity pc				-0.13	-0.12
				(0.25)	(0.26)
Left party in government					-0.01
					(0.01)
Year FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
				(con	tinued on next page)

Table A.1 (continued)

	(1)	(2)	(3)	(4)	(5)
Region FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Mean(y)	0.10	0.10	0.10	0.10	0.10
sd(y)	0.07	0.07	0.07	0.07	0.07
N.obs	238	238	238	238	238
R-squared	0.88	0.88	0.88	0.88	0.88

Note: The Table shows the results of regressing regional-level enrollment rates for the period 1987–2000 on a number of potential determinants. Institutional and financial factors are proxied by a dummy variable for the 1990 LOGSE reform, taking the value 1 for all years after 1990; regional public capital expenditure as % of GDP; per capita stock of public infrastructures devoted to education (Instituto Valenciano de Investigaciones Económicas, IVIE); a dummy for left party in government. Socio-economic factors are proxied by regional GDP per capita (Spanish Ministry of Finance and Public Administration); youth (16–19) unemployment rate; education level of active population, defined as the percentage of working-age population with primary education or less (IVIE). The expansion of the female employment rate is accounted for with data from the Economically Active Population Survey by INE (National Statistical Institute). As a proxy for crime rates back in time, data on regional Courts of Justice activity was retrieved from the General Attorney's Office of the Spanish Ministry of Justice. The correlation between provincial court data and provincial crime data obtained from the Spanish Ministry for Home Affairs show a correlation of 87.18% over the period 1993–1999. * p < 0.10, *** p < 0.05, *** p < 0.01. Several robustness checks and change of variables did not alter results.

Table A.2

Descriptive statistics.

	mean	sd	median
Crime actions /1000pax	29.69	160.02	0.00
Property c. actions /1000pax	13.05	99.81	0.00
Person c. actions /1000pax	9.68	83.25	0.00
Other c. actions /1000pax	6.95	64.82	0.00
Public preschool rate (0-3)	0.10	0.08	0.09
Mobility-adjusted public preschool rate (0–3)	0.10	0.08	0.09
Total preschool rate (0–3)	0.22	0.15	0.18
Public preschool rate (4–5)	0.62	0.12	0.60
Unemployed /1000pax	76.78	29.27	77.31
Police /1000pax	1.15	1.24	1.26
Municipal revenue pc (Thousands of E)	1.66	1.13	1.36
Municipal expenditure pc (Thousands of E)	1.63	1.12	1.35
Regional GDP pc at age 3 (Billions of E)	0.01	0.00	0.01
Regional unemployment (%) pc at age 3	18.30	5.61	18.02
Ν	153,924		

Table A.3

Descriptive statistics by crime typologies.

	Sample weight	Ν	Mean	sd
PROPERTY CRIMES	54.01%	153,924	13.69	102.15
Economic motivation	44.61%	153,924	9.87	86.59
Theft	23.80%			
Robbery	10.90%			
Car theft	2.50%			
Misappropriation	2.70%			
House occupation	2.40%			
Other property	2.30%			
Damages to property	4.87%	153,924	1.79	31.98
Fraud and falsity	4.54%	153,924	2.03	37.59
VIOLENT CRIMES	25.49%	153,924	8.65	77.98
Pure violence	18.71%	153,924	6.24	64.76
Injuries	12.50%			
Threats	6.20%			
Against the family	4.94%	153,924	1.75	32.55
Mistreatments	3.93%			
Missing family responsibilities	1.01%			
Gender violence	1.27%	153,924	0.50	16.62
Sexual crimes	0.57%	153,924	0.15	8.67
OTHER CRIMES	20.50%	153,924	7.09	65.58
Rules compliance	17.91%	153,924	6.48	62.37
Against road safety	10.11%			
Public order	6.09%			
Sentence break	1.00%			
Against public administration	0.72%			
Drugs	2.59%	153,924	0.61	18.31

Determinants of migration to Catalonia (2009-2014).

	(1)	(2)	(3)
Public preschool rate (0–3)	-0.04***	-0.00	-0.00
	(0.01)	(0.01)	(0.01)
GDP pc (age 3)			0.01
			(0.18)
Unemployment (age 3)			-0.00^{*}
			(0.00)
Year of birth FE		\checkmark	\checkmark
Region of birth FE		\checkmark	\checkmark
Year FE			\checkmark
Mean(y)	0.01	0.01	0.01
sd(y)	0.01	0.01	0.01
N.obs	1428	1428	1428
R-squared	0.22	0.86	0.87

Note: OLS regressions of the share of internal migrants from Spanish regions to Catalonia over the period 2009–2014 on potential factors influencing migration decisions. * p < 0.10, ** p < 0.05, *** p < 0.01.

Table A.5					
Characteristics of internal	migrants	by	preschool	exposure	quintile.

	1st Q		2nd Q		3rd Q		4th Q		5th Q	
	m	sd/SE	m	sd/SE	m	sd/SE	m	sd/SE	m	sd/SE
Education: LS		,		,		,		,		,
Migrants	0.25	(.01)	.25	(.01)	.24	(.01)	.31	(.01)	.29	(.01)
Stayers	.26	(.00)	.28	(.00)	.25	(.00)	.28	(.00)	.29	(.00)
Difference	01**	(.01)	04***	(.01)	-0.00	(.01)	.04	(.01)	.00	(.01)
Education: HS										
Migrants	.13	(.00)	.22	(.01)	.26	(.01)	.29	(.01)	.13	(.00)
Stayers	.13	(.00)	.25	(.00)	.24	(.00)	.27	(.00)	.16	(.00)
Difference	.00	(.00)	04***	(.01)	.03***	(.01)	.03***	(.01)	03***	(.00)
Education: Uni										
Migrants	.48	(.01)	.38	(.01)	.34	(.01)	.12	(.00)	.05	(.00)
Stayers	.48	(.00)	.32	(.00)	.33	(.00)	.13	(.00)	.06	(.00)
Difference	.00	(.01)	.06***	(.01)	.01**	(.01)	01***	(.00)	01***	(.00)
Active										
Migrants	.62	(.01)	.52	(.01)	.50	(.01)	.54	(.01)	.80	(.01)
Stayers	.61	(.00)	.47	(.00)	.56	(.00)	.69	(.00)	.83	(.00)
Difference	.01	(.01)	.05***	(.01)	06***	(.01)	15***	(.01)	03***	(.01)
Part-time										
Migrants	.21	(.01)	.26	(.01)	.30	(.01)	.37	(.02)	.31	(.03)
Stayers	.24	(.00)	.29	(.00)	.31	(.00)	.36	(.01)	.52	(.01)
Difference	03***	(.01)	03***	(.01)	00	(.01)	.01	(.02)	21***	(.03)

Note: For internal migrants and stayers, the table presents shares of individuals by highest education level achieved (Lower Secondary, Higher Secondary, University or equivalent), activity status (working or studying versus neither) and part-time work (yes or no) in each quintile of preschool exposure. Source: 2011 Census microdata. In parentheses, standard deviation of the shares and standard error of the differences between migrants and stayers.* p < 0.10, ** p < 0.05, *** p < 0.01. Figure A.5 shows this data graphically.

Table A.6

Preschool exposure and migrants' characteristics.

	8						
OUTCOME	Odds ratio	robust s.e.	Z	<i>p</i> -value	95% c.i.		Ν
2001 Census Education level 2011 Census	0.03	0.11	-0.82	0.41	0.00	169.46	1132
Education level Active Part-time	7.59 3.67 0.02	13.62 10.51 0.09	1.13 0.45 -0.86	0.26 0.65 0.39	.23 0.73 0.00	255.22 2.06 164.01	2063 1632 814

Note: The table indicates the relationship, in proportional odds ratios, between different characteristics of migrants and their preschool exposure at age 3, conditional on cohort and region of bith fixed effects. Ordered logit model estimates for education level achieved [Less than primary - Primary - Secondary - Specialization - University - Masters or higher]. Logit model estimates for being active (employed or studying) and for having part-time a job [Yes - No].

Characteristics of the crime age distributions.

	(1)	(2)	(3)	(4)	(5)
	Peak age	Low age	Peak / Low	sd	kurtosis
Public preschool rate (0-3)	-0.12	0.09	1.27	-0.73	0.81
	(0.14)	(0.29)	(2.49)	(2.14)	(0.66)
Year of birth FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Region of birth FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Mean(y)	21.05	20.76	4.10	41.02	2.21
sd(y)	3.29	5.25	3.31	46.50	0.70
N.obs	1512	1512	954	1512	1482
Pseudo R-squared	0.08	0.19	0.13	0.09	0.01
alpha	0.00	0.00	0.05***	0.28***	0.00

Note: SEs clustered (1-way) by cohort-region in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01.

Table A.8

Residence and birth region for 3-year olds across Spain.

	% difference		% difference
Andalusia	-0.18	Valencia	0.97
Aragon	-0.03	Extremadura	-1.16
Asturias	-1.16	Galicia	0.27
Balearic Islands	-0.30	Madrid	-0.64
Canary Islands	0.02	Murcia	-0.03
Cantabria	-0.00	Navarre	0.41
Castilla y Leon	-0.74	Basque Country	-1.44
Castilla-La Mancha	0.31	Rioja	1.30
Catalonia	-0.11	Ceuta and Melilla	-1.47

Note: For each Spanish region, the table shows the percentage difference between the number of 3-year old residents and the number of births recorded three years earlier (averages across birth cohorts 1984–1997). Source: municipal residence records, elaborated by the Spanish National Statistics Agency (INE).

Table A.9

Preschool exposure estimates under different robustness checks.

	Negative Binomial				
	(1)	(2)	(3)	(4)	(5)
C1) w/ regional controls until 12					-1.87**
					(0.77)
C2) w/ regional controls until 18					-2.20***
					(0.78)
C3) Social capital controls				-1.68***	-1.70**
				(0.63)	(0.67)
C4) Mobility-adjusted	-3.21***	-1.59***	-1.58***	-1.63**	-1.35*
	(0.67)	(0.61)	(0.59)	(0.64)	(0.75)
C5) Total enrolment	-1.86***	-1.27***	-1.24***	-1.35***	-1.36***
	(0.37)	(0.31)	(0.30)	(0.33)	(0.42)
C6) w/ exposure	-4.79***	-2.31***	-2.64^{***}	-2.49***	-1.89**
	(1.03)	(0.71)	(0.70)	(0.73)	(0.83)
C7) w/o islands and territories ^(x)	-3.17***	-1.62***	-1.56***	-1.57**	-1.20
	(0.68)	(0.62)	(0.60)	(0.65)	(0.79)
C8) w/o largest regions	-2.16***	-0.97	-0.95	-0.93	-0.93
	(0.62)	(1.41)	(1.44)	(1.67)	(1.64)
C9) w/o regions missing data	-3.08***	-1.68**	-1.66**	-1.72**	-1.43*
	(0.69)	(0.67)	(0.66)	(0.71)	(0.81)
C10) Province level (+)	-3.32***	-1.66	-1.90	-1.32	-1.17
	(1.14)	(1.75)	(1.85)	(1.85)	(1.85)
C11) Catalonia level (+)	-3.18***	-1.98	-2.03	-1.38	-1.11
	(0.66)	(1.83)	(1.84)	(1.99)	(1.86)
Local controls				Yes	Yes
Regional controls (at age 3)					Yes
Year of birth FE		\checkmark	\checkmark	\checkmark	\checkmark
Region of birth FE		\checkmark	\checkmark	\checkmark	\checkmark
Province FE			\checkmark	\checkmark	\checkmark
Year FE			\checkmark	\checkmark	\checkmark

Note: SEs clustered (1-way) by cohort-region in parentheses. * p < 0.10, ** p < 0.05, *** p < 0.01. Islands and territories are Canary and Balearic Islands, Ceuta and Melilla. (x) Largest regions are Catalonia and Andalusia. Regions missing some data points are Valencia and Galicia. (+) For regional level specifications, no local fixed effects are included. See descriptives for the provincial and regional level of aggregation in Table A.10.

Descriptive statistics at higher geographical aggregation .

	Provincial level			Regional level		
	mean	sd	median	mean	sd	median
Crime actions /1000pax	34.21	77.65	0	38.55	43.84	31.00
Unemployment /1000pax	61.16	10.90	63.88	63.45	5.69	64.03
Police /1000pax	1.64	0.74	1.73	1.69	0.33	1.76
Local revenue pc (Thousands of EUR)	1.64	0.38	1.58	1.52	0.14	1.50
Local expenditure pc (Thousands of EUR)	1.61	0.36	1.57	1.49	0.13	1.48
N	5801			1512		

Table A.11

Heterogeneous effects by relative economic advantage of birth cohorts.

	(1)	(2)	(3)	(4)	(5)	(6)
Public preschool rate (0–3)	-1.46**	-0.92	-1.74*	-1.30	-0.89	-0.08
	(0.68)	(0.89)	(0.96)	(1.07)	(1.19)	(1.30)
PP rate (0–3) x Low GDP	-0.39	-1.28				
	(1.02)	(1.08)				
PP rate $(0-3)$ x High unemployment			-0.07	-0.16		
			(0.88)	(0.92)		
PP rate $(0-3)$ x Hard crisis					-1.55	-2.15
					(1.35)	(1.41)
Local controls		Yes		Yes		Yes
Regional controls (at age 3)		Yes		Yes		Yes
Year of birth FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Region of birth FE	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark	\checkmark
Province FE		\checkmark		\checkmark		\checkmark
Year FE		\checkmark		\checkmark		\checkmark
Mean(y)	29.69	29.69	29.69	29.69	29.69	29.69
sd(y)	160.02	160.02	160.02	160.02	160.02	160.02
N.obs	153,924	153,924	153,924	153,924	153,924	153,924
R-squared						
Pseudo R-squared	0.00	0.00	0.00	0.00	0.00	0.00
alpha	35.84	35.24	35.84	35.24	35.84	35.24

Note: The table reports the results of interacting the preschool access measure with measures of relative economic advantage of birth cohorts. Low-GDP: equals 1 (0) for cohorts exposed to a GDP per capita below-median (above-median) within their region of origin at age 3. High-unemployment: equals 1 (0) for cohorts exposed to unemployment levels above-median (below-median) within their region of origin at age 3. Hard crisis: equals 1 (0) for cohorts who were aged 3 between 1993 and 1998 in regions hit hardest by the 1993 economic crisis (i.e. recording a rise in unemployment above the national average of +0.41% between 1991.T4 and 1994.T4). Each observation represents cohort 'c' born in region 'i' living in Catalan municipality 'm' in year 'y'. SEs clustered (1-way) by cohort-region in parentheses. * p < 0.10, *** p < 0.05, *** p < 0.01.

Table A.12

Descriptive statistics of empirical and predicted count distributions.

	Empirical distribution	Negative Binomial model	Poisson model
mean	.0001	.0001	.0001
standard deviation	.0263	.0261	.0047
skewness	31.5650	31.5313	5.6219
kurtosis	997.8949	996.4473	36.1710
max difference		.0230	.8306
value at max difference		0	1
mean difference		.0002	.0018
N	1001	1001	1001

Note: Comparison between the empirical offense counts distribution and the distributions of offense counts predicted by the Negative Binomial and the Poisson modeling options.

Table A.13

Bayesian and Akaike Information Criteria for the two modeling choices.

	BIC	AIC
Poisson model	$2.442 imes 10^7$	2.442×10^7
Negative Binomial model	4.684×10^5	$4.681 imes 10^5$
Difference	2.395×10^{7}	2.395×10^7
Prefer	NB model	NB model

Note: Assessment of model fit through Bayesian and Akaike Information Criteria for the Negative Binomial and Poisson modeling choices applied to our context.

Annual benefits in terms of crime reduction, by crime type.

Crime type	Total unit cost (ϵ)	Public unit cost (ϵ)	Multiplier	Annual total cost reduction (ϵ)	Annual public cost reduction (ϵ)
theft	1,942.1	900.7	81.1	10,317,486.2	4,784,921.1
robbery	15,930.6	7,669.8	57.2	27,474,533.2	13,227,580.0
car theft	14,481.1	5,629.2	33.4	3,284,053.8	1,276,600.1
damages	1,899.9	830.3	42.4	104,583.7	45,706.9
fraud	1,815.4	422.2	306.7	38,028,039.6	8,843,730.1
injury	19,772.6	4,813.0	6.4	16,345,683.4	3,978,806.9
threats	8,345.3	3,293.1	7.3	3,885,788.0	1,533,346.4
sexual	9,175.6	2,181.3	77.9	9,656,182.5	2,295,564.8
gender violence	9,175.6	2,181.3	63.5	4,845,458.5	1,151,911.1
TOTAL				113,941,808.8	37,138,167.6

Note: Total and public costs obtained from Heeks et al. (2018) and are expressed in 2015 EUR (exchange rate used GBP 1 = EUR 1.4073 as of 31/06/2015). Crime used in this exercise account for two thirds of crime with known offender recorded in Catalonia. The Multiplier inflates crime counts from those recorded by police with a known offender to an estimate of those actually occurred. The Multiplier is composed of a first inflation factor from crimes with known offender to crimes reported to police (calculated using administrative data for Catalonia, at our disposal), and a second inflation factor from crimes reported to police to total crimes (borrowed from Heeks et al., 2018). Annual estimates of cost reductions are obtained using impact estimates shown in Column 5 in Table 2 for the period 2009–2014, and are expressed in EUR discounted to 2004.

References

Abadie, A., Athey, S., Imbens, G.W., Wooldridge, J., 2017. When should you adjust standard errors for clustering? Technical Report. National Bureau of Economic Research.

Ackerman, J.M., Rossmo, D.K., 2015. How far to travel? A multilevel analysis of the residence-to-crime distance. J. Quant. Criminol. 31 (2), 237–262. doi:10. 1007/s10940-014-9232-7.

Almlund, M., Duckworth, A.L., Heckman, J.J., Kautz, T.D., 2011. Personality psychology and economics. In: Hanushek, E.A., Woessmann, L., Machin, S. (Eds.), Handbook of the Economics of Education. Elsevier, Amsterdam, pp. 1–181. doi:10.3386/w16822.

Anderson, D.M., 2014. In school and out of trouble? The minimum dropout age and juvenile crime. Rev. Econ. Stat. 96 (2), 318-331. Publisher: The MIT Press

Baker, M., Gruber, J., Milligan, K., 2008. Universal child care, maternal labor supply, and family well-being. J. Polit. Economy 116 (4), 709–745. Publisher: The University of Chicago Press

Baker, M., Gruber, J., Milligan, K., 2019. The long-run impacts of a universal child care program. Am. Econ. J. 11 (3), 1–26.

Becker, G.S., 1968. Crime and punishment: an economic approach. In: The Economic Dimensions of Crime. Springer, pp. 13-68.

Bell, B., Costa, R., Machin, S., 2016. Crime, compulsory schooling laws and education. Econ. Educ. Rev. 54, 214–226. doi:10.1016/j.econedurev.2015.09.007.

, 1992. Análisis de la demanda de servicios para la primera infancia. Estudios. Ministerio de Asuntos Sociales.

Berlinski, S., Galiani, S., Gertler, P., 2009. The effect of pre-primary education on primary school performance. J. Public Econ. 93 (1), 219–234. doi:10.1016/j. jpubeco.2008.09.002.

Blanden, J., Bono, E.D., McNally, S., Rabe, B., 2016. Universal pre-school education: the case of public funding with private provision. Econ. J. 126 (592), 682–723. doi:10.1111/ecoj.12374.

Bonal, X., Rambla, X., Calderón, E., Pros, N., 2005. La descentralización educativa en España: una mirada comparativa a los sistemas escolares de las Comunidades Autónomas. Estudis, vol. 18. Fundació Carles Pi i Sunyer.

Brownfield, D., Sorenson, A.M., 1993. Self-control and juvenile delinquency: theoretical issues and an empirical assessment of selected elements of a general theory of crime. Deviant Behav. 14 (3), 243–264.

Buonanno, P., Leonida, L., 2009. Non-market effects of education on crime: evidence from Italian regions. Econ. Educ. Rev. 28 (1), 11-17.

Burchinal, M., Vandergrift, N., Pianta, R., Mashburn, A., 2010. Threshold analysis of association between child care quality and child outcomes for low-income children in pre-kindergarten programs. Early Child Res. O. 25 (2), 166–176. doi:10.1016/j.ecresq.2009.10.004.

Burt, C.H., Simons, R.L., Simons, L.G., 2006. A longitudinal test of the effects of parenting and the stability of self-control: negative evidence for the general theory of crime. Criminology 44 (2), 353–396.

Burton Jr, V.S., Cullen, F.T., Evans, T.D., Alarid, L.F., Dunaway, R.G., 1998. Gender, self-control, and crime. J. Res. Crime Delinquency 35 (2), 123-147.

Calero, J., Bonal, X., 1999. Política educativa y gasto público en educación. aspectos teóricos y una aplicación al caso español. Pomares-Corredor, Barcelona. Calero, J., Bonal, X., 2004. Financiación de la educación en España. In: Navarro, V. (Ed.), El Estado del Bienestar en España. Tecno-UPF, Madrid.

Cameron, A.C., Trivedi, P.K., 2010. Microeconometrics using stata, Revised Ed. Stata Press.

Campbell, F.A., Conti, G., Heckman, J.J., Moon, S.H., Pinto, R., Pungello, E., Pan, Y., 2014. Early childhood investments substantially boost adult health. Science 343 (6178), 1478–1485. doi:10.1126/science.1248429.

Campbell, F.A., Ramey, C.T., Pungello, E., Sparling, J., Miller-Johnson, S., 2002. Early childhood education: young adult outcomes from the abecedarian project. Appl. Dev. Sci. 6 (1), 42–57. doi:10.1207/S1532480XADS0601_05.

Campbell, F.A., Wasik, B.H., Pungello, E., Burchinal, M., Barbarin, O., Kainz, K., Sparling, J.J., Ramey, C.T., 2008. Young adult outcomes of the Abecedarian and CARE early childhood educational interventions. Early Child Res. Q. 23 (4), 452–466.

Card, D., Krueger, A.B., 1992. Does school quality matter? Returns to education and the characteristics of public schools in the United States. J. Polit. Economy 100 (1), 1–40. doi:10.1086/261805.

Carneiro, P.M., Heckman, J.J., 2003. Human Capital Policy. IZA Discussion Paper No. 821.

Carta, F., Rizzica, L., 2018. Early kindergarten, maternal labor supply and children's outcomes: evidence from Italy. J. Public Econ. 158, 79-102.

Case, A., Lubotsky, D., Paxson, C., 2002. Economic status and health in childhood: the origins of the gradient. Am. Econ. Rev. 92 (5), 1308-1334. doi:10. 1257/000282802762024520.

Cauffman, E., Steinberg, L., Piquero, A.R., 2005. Psychological, neuropsychological and physiological correlates of serious antisocial behavior in adolescence: the role of self-control. Criminology 43 (1), 133–176.

Conti, G., Heckman, J.J., 2014. Economics of child well-being. In: Handbook of Child Well-Being. Springer, pp. 363–401.

Conti, G., Heckman, J.J., Pinto, R., 2016. The effects of two influential early childhood interventions on health and healthy behaviour. Econ. J. 126 (596). Cornelissen, T., Dustmann, C., Schönberg, U., Raute, A., 2017. Who benefits from universal child care? : estimating marginal returns to early child care attendance. J. Polit. Economy.

Currie, J., Thomas, D., 1995. Does head start make a difference? Am. Econ. Rev. 85 (3), 341-364.

Del Campo Urbano, S., 1982. La evolución de la familia española en el siglo XX. Alianza Editorial, Madrid.

Deming, D., 2009. Early childhood intervention and life-cycle skill development: evidence from head start. Am. Econ. J. 1 (3), 111-134.

Deming, D.J., 2011. Better schools, less crime? O. J. Econ. 126 (4), 2063-2115.

Duflo, E., 2001. Schooling and labor market consequences of school construction in Indonesia: evidence from an unusual policy experiment. Am. Econ. Rev. 91 (4), 795-813.

Dumas, C., Lefranc, A., 2012. Early schooling and later outcomes: evidence from pre-school extension in France. In: Ermisch, J., Jantti, M., Smeeding, T. (Eds.), Inequality from Childhood to Adulthood: A Cross-National Perspective on the Transmission of Advantage. New York, Russell sage foundation

Duncan, G.J., Magnuson, K.A., 2013. Investing in preschool programs. J. Econ. Perspect. 27 (2), 109-132. doi:10.1257/jep.27.2.109.

Duncan, G.J., Ziol-Guest, K.M., Kalil, A., 2010. Early-childhood poverty and adult attainment, behavior, and health. Child Dev. 81 (1), 306-325. doi:10.1111/j. 1467-8624.2009.01396.x.

Felfe, C., Lalive, R., 2018. Does early child care affect children's development? J. Public Econ. 159, 33-53.

Felfe, C., Nollenberger, N., Rodríguez-Planas, N., 2015. Can't buy mommy's love? universal childcare and children's long-term cognitive development. J. Popul. Econ. 28 (2), 393-422.

Figlio, D.N., Roth, J., 2009. The behavioral consequences of pre-kindergarten participation for disadvantaged youth. In: Gruber, J. (Ed.), The Problems of Disadvantaged Youth: An Economic Perspective. University of Chicago Press, National Bureau of Economic Research, pp. 15-42

Flaguer, L., Oliver, E., 2002, Polítiques de suport a les famílies. In: Gómez-Granell, C., García-Milá, M., Ripol-Millet, A., Panchón, C. (Eds.), La infáncia i les families als inicis del segle XXI, vol. 1. Institut d'Infáncia i Món Urbá. Observatori de la infáncia i la família., Barcelona, pp. 299–349. Fort, M., Ichino, A., Zanella, G., 2020. Cognitive and noncognitive costs of day care at age 0–2 for children in advantaged families. J. Polit. Economy 128 (1),

158-205. doi:10.1086/704075. Publisher: The University of Chicago Press

Garces, E., Thomas, D., Currie, J., 2002. Longer-term effects of head start. Am. Econ. Rev. 92 (4), 999-1012. doi:10.1257/00028280260344560.

Gertler, P., Heckman, J.J., Pinto, R., Zanolini, A., Vermeersch, C., Walker, S., Chang, S.M., Grantham-McGregor, S., 2014. Labor market returns to an early childhood stimulation intervention in Jamaica. Science 344 (6187), 998-1001. doi:10.1126/science.1251178.

González, M.J., 2004. La escolarización de la primera infancia en España: desequilibrios territoriales y socioeconómicos en el acceso a los servicios. In: Navarro, V. (Ed.), El Estado del Bienestar en España. Tecnos, Madrid, pp. 233-291.

Gonzälez, M.J., Ouiroga, 2003. Per qué no hi ha escoles bressol públiques al meu municipi? L'escolarització de la primera infáncia a Catalunya en els contextos europeu i espanyol. In: Navarro, V. (Ed.), L'Estat del Benestar a Catalunya. Diputació de Barcelona, Barcelona.

González, M.J., Vidal Torre, S., 2005. Where do I leave my baby? Use and Development of Early Childcare in Spain. DemoSoc Working Paper 02.

González, M.J., Vidal Torre, S., 2006. Où vais-je laisser mom bébé? utilisation et developpement des structures de garde d'enfants en Espagne. Recherches et Prévisions 83, 233-291.

Groot, W., van den Brink, H.M., 2010. The effects of education on crime. Appl. Econ. 42 (3), 279-289.

Gunnar, M.R., Barr, R.G., 1998. Stress, early brain development, and behavior. Infants Young Children 11 (1), 1-14.

Gupta, N.D., Simonsen, M., 2010. Non-cognitive child outcomes and universal high quality child care. J. Public Econ. 94 (1-2), 30-43.

Guryan, J., Hurst, E., Kearney, M., 2008. Parental education and parental time with children. J. Econ. Perspect. 22 (3), 23-46. doi:10.1257/jep.22.3.23.

Hahn, R., Fuqua-Whitley, D., Wethington, H., Lowy, J., Crosby, A., Fullilove, M., Johnson, R., Liberman, A., Moscicki, E., Price, L., Snyder, S., Tuma, F., Cory, S., Stone, G., Mukhopadhaya, K., Chattopadhyay, S., Dahlberg, L., Task Force on Community Preventive Services, 2007. Effectiveness of universal school-based programs to prevent violent and aggressive behavior: a systematic review. Am. J. Prev. Med. 33 (2 Suppl), S114-129. doi:10.1016/j.amepre.2007.04.012.

Havnes, T., Mogstad, M., 2011. Money for nothing? Universal child care and maternal employment. J. Public Econ. 95 (11-12), 1455-1465. Publisher: Elsevier Havnes, T., Mogstad, M., 2011. No child left behind: subsidized child care and children's long-run outcomes. Am. Econ. J. 3 (2), 97-129. doi:10.1257/pol.3.2.97.

- Havnes, T., Mogstad, M., 2015. Is universal child care leveling the playing field? J. Public Econ. 127, 100-114.
- Heckman, J.J., 2000. Policies to foster human capital. Res. Econ. 54 (1), 3–56.

Heckman, J.J., 2006. Skill formation and the economics of investing in disadvantaged children. Science 312 (5782), 1900–1902. doi:10.1126/science.1128898. Heckman, J.J., Malofeeva, L., Pinto, R., Savelyev, P.A., 2009. The powerful role of noncognitive capabilities in explaining the effects of the perry preschool program. Unpublished Manuscript. University of Chicago, Department of Economics.

Heckman, J.J., Moon, S.H., Pinto, R., Savelyev, P.A., Yavitz, A., 2009. A reanalysis of the highscope perry preschool program. Unpublished Manuscript. University of Chicago, Department of Economics.

Heckman, J.J., Moon, S.H., Pinto, R., Savelyev, P.A., Yavitz, A., 2010. Analyzing social experiments as implemented: a reexamination of the evidence from the HighScope Perry Preschool program. Quant. Econom. 1 (1), 1-46.

Heckman, J.J., Moon, S.H., Pinto, R., Savelyev, P.A., Yavitz, A., 2010. The rate of return to the HighScope Perry Preschool Program. J. Public Econ. 94 (1-2), 114-128.

Heckman, J.J., Pinto, R., Savelyev, P.A., 2013. Understanding the mechanisms through which an influential early childhood program boosted adult outcomes. Am. Econ. Rev. 103 (6), 2052-2086.

Heeks, M., Reed, S., Tafsiri, M., Prince, S., 2018. The economic and social costs of crime second edition. Research Report 99. UK Home Office, London, UK.

Hidalgo-Hidalgo, M., García-Pérez, J.L., 2012. Impacto de la asistencia a Educación Infantil sobre los resultados académicos del estudiante en Primaria PIRLS 2011 International Results in Reading, Amsterdam, the Netherlands: TIMSS & PIRLS International Study Center.

Hilbe, J.M., 2011. Negative binomial regression, second ed. Cambridge University Press, Cambridge, UK; New York.

Hirschi, T., 1969. Causes of delinquency. University of California Press.

Hovnes, H., Page, M., Stevens, A.H., 2011, Can targeted transfers improve birth outcomes?: evidence from the introduction of the WIC program. I. Public Econ. 95 (7-8), 813-827. Publisher: Elsevier

Johnson, R.C., Jackson, C.K., 2019. Reducing inequality through dynamic complementarity: evidence from head start and public school spending. Am. Econ. J. 11 (4), 310-349.

Junger, M., Tremblay, R.E., 1999. Self-control, accidents, and crime. Crim. Justice Behav. 26 (4), 485-501.

Kagan, J., Snidman, N., 1999. Early childhood predictors of adult anxiety disorders. Biol. Psychiatry 46 (11), 1536–1541.

Keane, C., Maxim, P.S., Teevan, J.J., 1993. Drinking and driving, self-control, and gender: testing a general theory of crime. J. Res. Crime Delinquency 30 (1), 30-46

Kimmel, J., Connelly, R., 2007. Mothers' time choices caregiving, leisure, home production, and paid work. J. Hum. Resour. 42 (3), 643-681.

Kleiber, C., Zeileis, A., 2016. Visualizing count data regressions using rootograms. Am. Stat. 70 (3), 296-303.

Kottelenberg, M.J., Lehrer, S.F., 2013. New evidence on the impacts of access to and attending universal child-care in Canada. Can. Public Policy 39 (2), 263-286. Publisher: University of Toronto Press

Le, A.T., Miller, P.W., Heath, A.C., Martin, N., 2005. Early childhood behaviours, schooling and labour market outcomes: estimates from a sample of twins. Econ. Educ. Rev. 24 (1), 1-17. doi:10.1016/j.econedurev.2004.04.004

Ley Orgánica, 1990. 1/1990, de 3 de octubre, de Ordenación General del Sistema Educativo.

Lochner, L., Moretti, E., 2004. The effect of education on crime: evidence from prison inmates, arrests, and self-reports. Am. Econ. Rev. 94 (1), 155-189. Loeb, S., Bridges, M., Bassok, D., Fuller, B., Rumberger, R.W., 2007. How much is too much? The influence of preschool centers on children's social and cognitive development. Econ. Educ. Rev. 26 (1), 52-66. doi:10.1016/j.econedurev.2005.11.005.

Long, J.S., Freese, J., 2014. Regression models for categorical dependent variables using stata, third ed. Stata Press.

Ludwig, J., Miller, D.L., 2007. Does head start improve children's life chances? evidence from a regression discontinuity design. Q. J. Econ. 122 (1), 159-208. Machin, S., Marie, O., Vujic, S., 2011. The crime reducing effect of education. Econ. J. 121 (552), 463-484.

Machin, S., Vujic, S., Marie, O., 2012. Youth crime and education expansion. German Econ. Rev. 13 (4), 366-384.

Magnuson, K.A., Ruhm, C., Waldfogel, J., 2007. Does prekindergarten improve school preparation and performance? Econ. Educ. Rev. 26 (1), 33-51. doi:10. 1016/i.econedurev.2005.09.008.

Maslow, A.H., 1943. A theory of human motivation. Psychol. Rev. 50 (4), 370.

- Moffitt, T.E., Arseneault, L., Belsky, D., Dickson, N., Hancox, R.J., Harrington, H., Houts, R., Poulton, R., Roberts, B.W., Ross, S., Sears, M.R., Thomson, W.M., Caspi, A., 2011. A gradient of childhood self-control predicts health, wealth, and public safety. Proc. Natl. Acad. Sci. 108 (7), 2693–2698. doi:10.1073/ pnas.1010076108.
- Nelson, C.A., Sheridan, M.A., 2011. Lessons from neuroscience research for understanding causal links between family and neighborhood characteristics and educational outcomes. In: Duncan, G.J., Murnane, R.J. (Eds.), Whither opportunity: Rising Inequality, Schools, and Children's Life Chances. Russell Sage, New York, pp. 27–46.
- Nguyen, T., Watts, T.W., Duncan, G.J., Clements, D.H., Sarama, J.S., Wolfe, C., Spitler, M.E., 2016. Which preschool mathematics competencies are most predictive of fifth grade achievement? Early Child Res. Q. 36, 550–560. doi:10.1016/j.ecresq.2016.02.003.

Phillips, D.A., Shonkoff, J.P., 2000. From neurons to neighborhoods: the science of early childhood development. National Academies Press.

- Rambla, X., Bonal, X., 2000. La política educativa y la estructura social. In: Adelantado, J. (Ed.), Cambios en el estado del bienestar. Políticas sociales y desigualdades en España. Icaria, Barcelona.
- Robinson, M., 2007. Child development from birth to eight: a journey through the early years: a journey through the early years. McGraw-Hill Education (UK).

Rolnick, A., Grunewald, R., 2003. Early childhood development: economic development with a high public return. Region 17 (4), 6-12.

- Romero-Valiente, J.M., 2003. Migraciones. In: Arroyo Pérez, A. (Ed.), Tendencias demográficas durante el siglo XX en España. Instituto Nacional de Estadistica, España.
- Santín, D., Sicilia, G., 2015. El impacto de la educación infantil en los resultados de primaria: evidencia para España a partir de un experimento natural. Reflexiones sobre el sistema educativo español 45.
- Schweinhart, L.J., Montie, J., Xiang, Z., Barnett, W.S., Belfield, C.R., Nores, M., 2005. Lifetime effects: the high/scope perry preschool study through age 40. High/Scope Foundation Press, 14. Ypsilanti, MI.

Smith, A., 2015. The long-run effects of universal pre-k on criminal activity. SSRN Working Paper 2685507 doi:10.2139/ssrn.2685507.

- Tarrach, A., Serra, A., Baños, J., 2011. Pautes per al cálcul i l'análisi de costos de l'escola bressol municipal. Technical Report. Diputació Barcelona, Barcelona. Temporeros y educación, 1997. la atención educativa a los hijos de trabajadores temporeros. Technical Report. Defensor del Pueblo Andaluz, Andalucía.
- Thompson, O., 2018. Head start's long-run impact evidence from the programs introduction. J. Hum. Resour. 53 (4), 1100–1139. Publisher: University of Wisconsin Press
- Tremblay, R.E., Nagin, D.S., Séguin, J.R., Zoccolillo, M., Zelazo, P.D., Boivin, M., Pérusse, D., Japel, C., 2004. Physical aggression during early childhood: trajectories and predictors. Pediatrics 114 (1), e43–e50. doi:10.1542/peds.114.1.e43.
- Tukey, J.W., 1972. Some graphic and semigraphic displays. In: Statistical Papers in Honor of George W. Snedecor. Iowa State University Press, Ames, IA, pp. 293–316. Reprinted in William S. Cleveland (ed.): The Collected Works of John W. Tukey, Volume V. Graphics: 1965–1985, Wadsworth & Brooks/Cole, Pacific Grove, CA, 1988
- Tukey, J.W., 1977. Exploratory data analysis. Addison-Wesley, Reading, MA.
- Wainer, H., 1974. The suspended rootogram and other visual displays: an empirical validation. Am Stat 28 (4), 143-145. doi:10.1080/00031305.1974. 10479098.
- White, J.L., Moffitt, T.E., Caspi, A., Bartusch, D.J., Needles, D.J., Stouthamer-Loeber, M., 1994. Measuring impulsivity and examining its relationship to delinquency. J. Abnorm. Psychol. 103 (2), 192–205. doi:10.1037/0021-843X.103.2.192.
- Wikström, P.-O.H., Svensson, R., 2010. When does self-control matter? the interaction between morality and self-control in crime causation. Eur. J. Criminol. 7 (5), 395-410.
- Wikström, P.-O.H., Treiber, K., 2007. The role of self-control in crime causation: beyond gottfredson and Hirschi's general theory of crime. Eur. J. Criminol. 4 (2), 237-264.